

Managing cross-cultural negotiations in the current Northern Iraq International Business environment:

An exploratory study.

Tobias Blechschmidt Master of Business Managing cross-cultural negotiations in the current Northern Iraq International Business environment: An exploratory study.

#### **Tobias Blechschmidt**

A thesis submitted to
Auckland University of Technology
in partial fulfilment of requirement for the degree of
Master of Business

November 2013 Faculty of Business and Law Supervisor: Dr, Sabina Jaeger, PhD

### **Table of Contents**

Table of Contentsi	
List of Tables	iii
Acknowledgements	v
Ethics Approval	<b>V</b> i
Abstract	vii
Introduction	1
Background of the Study	1
The Land and People of Modern Iraq	
Latest developments – economic, social and cultural change in northern Iraq after 2003	
Aims of This Research	
Organisation of the Study	
Literature Review	
Introduction	8
Negotiation Goal	9
Negotiation Attitude	10
Communication	12
View of Time	14
Emotionalism	15
Risk Propensity	17
Summary	19
Research Design	20
Research aim	20
Philosophical Approach	21
Data Collection	23
The interview questions	24
Interview process	26
Secondary data	27
Participant sampling	27
Sample size and demographics	28
Analytic Approach	29
Limitations	31

Ethical Considerations	31
Summary	32
Results and Discussion	33
Theme 1. Goal of Business Negotiations: When Negotiating What Are Your Goals?	34
Theme 2. Attitude Towards Negotiation: What is Your Attitude Towards Negotiations?	37
Theme 3. Methods of Communication: During Your Negotiations, How Do You Communicate With Your Opposite?	39
Theme 4. Sensitivity to Time: What Value Does Time Have For You When Negotiating?	41
Theme 5. Emotionalism: When Negotiating What Is Your Opinion of Emotions at The Negotiation Table?	44
Theme 6. Willingness to Take Risks: Do You Feel Comfortable in Taking Risks?	47
Summary of Findings	49
Conclusions	51
Limitations of the Study	53
Implications and Recommendations for Future Research	54
Reference List	56
Appendices	73

### List of tables

Table 1: Interview questions	. 24
Table 2: Details of the participants	8
Table 3: Phases of grounded theory	29
Table 4: Themes identified through analysis of the research data	33

# Attestation of authorship

I hereby declare that this submission is my own work and that, to the
best of my knowledge and belief, it contains no material previously
published or written by another person (except where explicitly defined
in the acknowledgements), nor material which to a substantial extent has
been submitted for the award of any other degree or diploma of a university
or other institution of higher learning.
AUTHOR SIGNATURE:

DATE:

#### Acknowledgements

This research project would not have been possible without the support of many people. The author wishes to express his gratitude to his supervisor, Dr Sabina Jaeger, Lecturer, Department of Business and Law, Auckland University of Technology who was abundantly helpful and offered invaluable assistance, support and guidance. The author wishes to express his love and gratitude to his beloved family; for their understanding and endless love, through the duration of his studies.

## **Ethics Approval**

Ethics approval for this research was granted by the Auckland University of Technology Ethics Committee on 24th May 2013, approval number **13/94**.

# Managing Cross-Cultural Negotiations in the Current Northern International Business Environment

#### **Abstract**

This study investigates factors that impact cross-cultural negotiations in the current international business environment of northern Iraq. In spite of considerable political volatility between the Kurdish regional government and the central government in Baghdad the region has attracted significant attention from Western companies targeting its rich oil resources. Therefore the focus of this small, qualitative study is on cross-cultural negotiations in the current IB environment. Eight in-depth interviews with negotiators from Western and Kurdish-Iraqi backgrounds were conducted to explore the differences in negotiation practices in this interesting region. The data from these in-depth and semi-structured interviews were then analysed through a modified grounded theory approach. The findings of this study showed specific behaviour patterns which are discussed in relation to the extant literature on intercultural negotiation. Major points of difference in the negotiating styles are the differing time expectations, motivational goals, communication approach and emotionalism. The variations in the underlying cultural concepts are explicit in the area of cross-cultural negotiation. These disparities are reflected in the negotiation goals and attitudes which either favour higher economic outcomes or the development of more general positive business relationships. Considering the limited number of studies on cross-cultural negotiations in Iraq, the identified negotiation patterns provide valuable insights into the local intercultural negotiation practices. Knowing these differences in negotiation behaviour shapes expectations of Western and local negotiators in the Kurdistan region. This initial research project should be seen as a first step in the attempt to expand the knowledge of negotiation practices in the area.

#### Chapter 1

#### Introduction

In an increasingly global business environment the success of international business relationships depends on effective business negotiations. Managers must interact effectively with people from different cultural backgrounds and who have different values, behavioural norms and ways of perceiving reality (Adler, 2002). Many misunderstandings and breakdowns in intercultural negotiations result from expectations about the negotiations that were not shared by all parties involved (Adler, 1986). Differences in culture can influence the way people from other cultures conduct themselves in negotiating sessions and how they conceive the very nature and function of negotiation itself (Weiss, 1994). In other words, in international business, cross-cultural negotiations harbour cultural differences which are inevitable between negotiators from different backgrounds. This diversity of values leads to different approaches and styles used in the negotiation process. Most individuals believe that the other negotiator shares their "common sense" assumptions which, to them, are familiar and unquestioned when negotiating inside their own culture. There is also the general perception that when their counterparts act unexpectedly in international negotiations that they are less committed (Kimmel, 1994). Therefore, to develop a less presumptive approach to intercultural negotiations one requires insights into the subjective cultures and communication habits that foreign individuals acquire as a result of their socialisation (Kimmel, 1994.) One has better success in cross-cultural negotiations by understanding the other negotiating party and so it is important to learn about intercultural encounters and recognise their differences.

#### **Background of the Study**

When US forces invaded Iraq in 2003 they found, among other issues, a poor infrastructure in an oil industry that produced about one million barrels per day. This was only a fraction of the country's output in the late 1980s (Walt, 2013). The country was bankrupt due to decades of sanctions and fighting. This was astonishing

in a resource-rich country with a proven 143.1 billion barrels of oil reserves, the fifth largest reserve in the world. Now, more than a decade later, the economic situation in the country is improving. In 2012 Iraq produced more than three million barrels per day which was their highest level of production in decades (Walt, 2013). The International Energy Agency (IEA) estimates in their report entitled *Iraq Energy Outlook*, that the daily output could rise to 8.3 million barrels per day in 2035 and exceed the current production levels of Saudi Arabia (IEA, 2012). In addition the IEA expects that Iraq is highly likely to become the biggest contributor to world oil supplies and this has the potential to transform the country into a major power in global trade (IEA, 2012).

Recent discoveries of natural oil resources that are said to exceed predictions sparked interest in the economy of the semi-autonomous region of Kurdistan-Iraq (The Economist, 2013). The Kurdistan regional government (KRG) estimates that the oil reserves in their territory at about 45 billion barrels of oil along with about 10 trillion cubic feet of natural gas, with the number of new discoveries being increasingly rare in the world (Walt, 2013). This potential has led to more than 50 international companies, including Exxon Mobil, Chevron, Hess, Marathon, HKN Energy, Gazprom Neft, Total and several Turkish companies, to sign deals with the KRG. The KRG offers more favourable production-sharing agreements than the central government in Baghdad. The region has its own president, prime minister and parliament. The population speaks Kurdish but they are nonetheless still part of Iraq (Walt, 2013). According to the KRG the GDP growth for the economy is likely to increase by eight per cent per year till 2016. Thus, there are good reasons for Western companies to invest in Kurdistan. This is so despite the escalating tensions with the central government that suspects that the KRG is trying to use their newfound oil wealth to create a breakaway state (Walt, 2013).

Recent attempts by the KRG to gain their sovereignty and create a Kurdish state separate from the central government in Baghdad heightens the need to understand how the population in the Kurdistan region differentiates itself from the rest of the country. Considering the differences in culture between the Kurdish and Arab population of Iraq it is important to investigate the cultural factors that influence intercultural negotiations in this region. This study undertakes to examine the

different factors that can be observed in intercultural negotiations in the oil industry of northern Iraq. The aim is to identify patterns of specific actions that enable international managers to successfully conduct negotiations in this region. This study might help to stimulate further research in order to develop effective strategies and tactics to operate in the Kurdish business environment and to make negotiations more successful.

#### The Land and People of Modern Iraq

The state of Iraq is a creation of the twentieth century, brought into being by politicians and external forces. The area of Iraq includes a few semi-autonomous areas and is home to several diverse cultural groups that all shape the identity of contemporary Iraq. This cultural diversity has been a strength as well as a challenge in harnessing Iraq's rich resources and the mixture of its inhabitants has been a preoccupation of Iraq's leaders in the past as well as in the present (Marr, 2012).

Marr (2012) has suggested that, although Iraq is internationally recognised as a state, it is not possible to speak of an Iraqi nation. The diverse population forms neither a single community in a political sense nor do they have a common sense of identity yet. The different ethnicities cause a fundamental demographic divide which is obvious in the different parts of the population. Arabic speakers constitute of 75 to 80 per cent of the population; Kurdish speakers make up 15 to 20 per cent (Marr, 2012). The Arabs dominate the western desert and the valleys of the Tigris and Euphrates from Basra to the Mosul plain, and the Kurds have their territory in the mountain terrain of the north and east and the foothills that adjoin it. It is estimated that in 2011 the Kurdish population of Iraq was just over 6 million. However, the Iraqi Kurds are only one part of the entire Kurdish population with whom they identify on linguistic, cultural and nationalistic grounds (Marr, 2012). So far, the Kurdish population does not have a state of their own and is living in various countries as an ethnic diaspora, a situation where they suffer considerable discrimination. Other areas with a Kurdish population include Turkey – 13 million, or 19 per cent of Turkey's population – nine million in Iran, two million in Syria and smaller numbers in Armenia, Azerbaijan, and Europe (Marr, 2012). The population in Iraq is also divided along religious lines between the two dominant sects of Islam, Shi'a and Sunni. However, as the majority of Kurdish people are Sunni, this division mostly impacts the Arabs; religious differences have segmented into three distinct communities: Arab Shi'a, Arab Sunnis and the Kurds (Marr, 2012).

Over centuries Shi'a Islam has had its stronghold in southern Iraq, as the Arab tribes migrated from the Arabian Peninsula during the eighteenth and nineteenth centuries. Today the Shi'a are the largest single group in Iraq, even outnumbering the Arab Sunnis by a ratio of three to one, and constitute s the majority group in the total population (Marr, 2012). During the Sunni Ottoman administration in the sixteenth century, the Iraqi Shi'a were excluded from administrative positions, the military and the educational institutions. As a result, over the course of time, the Shi'a became alienated from the Sunni governments (Marr, 2012).

By contrast, the Arab Sunnis in Iraq tend to be more secular and more urbanised than their Shi'a counterparts and their communal identity is less developed. But, despite their minority status, the Arab Sunnis have dominated the political and social life of Iraq since the sixteenth century (Marr, 2012). It is estimated that the Arab Sunnis account for 15 to 20 per cent of the population. They are almost entirely situated in the cities and towns of the central and Northern provinces (Marr, 2012). Bigger numbers of Sunnis also settled in some selected cities in the south including Basra. The displacement of the Sunni government in 2003 by a new Shi'a leadership was a social and political change of major proportions and sparked hostilities between the two sects (Marr, 2012).

The Kurds, due to their numbers, geographic concentration, cultural and linguistic identity, have always proved the most difficult of Iraq's people to assimilate (Marr, 2012). The Kurdish people speak an Indo-European language similar to Persian. Their development in the twentieth century was based on language, close tribal ties, customs and a shared history inspired by Kurdish nationalist movements. The majority of the Kurdish population is centred in Erbil, its political hub, and the intellectual centre of Sulaymaniya (Marr, 2012). Local wars and actions taken by the Iraqi government under Saddam Hussein forced a migration of the Kurdish population and resulted in the resettlement of the large, country population in cities and towns (Marr, 2012). Aside from the three major demographic groups there are

also several smaller ethnic and religious groups such as Turkmen, Shi'i Persians, Shi'i Kurds, Jews and Christians (Chaldeans, Assyrians, Armenian, Jacobites, Greek Orthodox, Greek, Catholic and Latin Catholic) (Marr, 2012)

# Latest developments – economic, social and cultural change in northern Iraq after 2003

Before 2003, Iraq's political, economic and social structure had deteriorated gradually under the regime of Saddam Hussein but the American intervention brought about a complete and immediate political and social collapse. As there was no clear direction for the future, the dismantling of the previous political and military structures led to the creation of a political and social vacuum. Consequently, Iraq soon began to fracture into its different ethnic and sectarian components (Marr, 2012).

While the devastation affected Arab areas, the Kurds strengthened their position in the north. Between 2006 and 2010, the Kurds used their well-developed security apparatus sealing off their area and maintaining a zone of relative peace. In this newfound "Iraqi Kurdistan" political leaders expanded the economy and undertook the development of their political, social and cultural institutions (Marr, 2012). The leaders of Kurdistan deliberately set their region apart from the rest of Iraq and the new Iraqi constitution. This included that the KRG recognized as a region with legal authority.

The discoveries of new oil and gas resources in the three provinces were recognised by the Iraqi constitution as constituting the KRG (Marr, 2012). By September 2008, the KRG had signed over twenty foreign oil and gas contracts. Consequently a dispute with the central government arose over the issue of who has control over the oil resources and whether the KRG has the right to conclude independent oil contracts in its territory. The KRG, driven by the desire to develop an independent oil sector in their territory, had a different attitude towards foreign direct investment. The KRG was more "market friendly" than the central government and was willing to entice international companies with better terms and conditions. After several attempts to reconcile these differences the central government refused to legally

recognise any contracts concluded by the KRG (Marr, 2012). In addition, the central government announced that it would bar any international company with a contract in the north from bidding for their oil fields in the south. In June 2007, the central government and the KRG agreed on a revenue-sharing deal and the central government established a fund for oil revenues, automatically transferring 17 per cent of the revenues to the KRG.

By 2009, economic dependency on the central government had become a dominant issue for the KRG and achieving not only political and military but economic independence had become an important goal. Nevertheless, oil was not the only economic front the KRG advanced and made vigorous efforts to attract direct foreign investments and succeeded in a number of areas. Turkey, Iran and the United Arab Emirates responded, among others, and the Kurds were also assured by the central government of 17 per cent of the national Iraqi budget (Marr, 2009). Kurdish development expenditure went into infrastructure, education, health and services, attempting to establish a northern social and cultural capital.

Meanwhile, the Kurds also continued their efforts to establish an independent identity for their region and underpin it with cultural and historical foundations (Marr, 2012). In order to set itself apart, the KRG altered the school curriculum and substituted Kurdish history and culture for the Arabic content which was compulsory under the regime of Saddam Hussein. This created a younger generation of Kurds that is essentially unable to read or speak Arabic which is the official language in the rest of Iraq (Marr, 2012). The growing ethnic divide between the Kurds and the Arab areas of the country, based mainly on language and growing institutional separation in the north, will make the solution of economic and political problems with the central government more difficult in the future.

#### **Aims of This Research**

The aim of this study is to explore how cross-cultural negotiations between Western and Kurdish-Iraqi senior managers are conducted in the international business environment of northern Iraq. In order to attain this goal, the current research focuses on one objective. This objective is to investigate and explore the six factors of

cultural difference that normally impact cross-cultural negotiations elsewhere. These differences will be explored in the social interactions between Kurdish-Iraqi and Western senior managers.

The study involves interviews with eight senior managers from various backgrounds who have experience with cross-cultural negotiations due to their jobs and positions. These interviews will be transcribed and grounded theory and coding technique will be used to analyse the data.

Given the limited amount of empirical research in the field of intercultural negotiation in Iraq, the contribution of this study lies in the gathering of empirical data and subsequent knowledge creation. The study should provide valuable insights into the local intercultural negotiation practices. The implications are of concern to Western and local negotiators when shaping expectations about the differences in negotiation approaches.

#### **Organisation of the Study**

This dissertation consists of five chapters. The first chapter outlines the focus of this study by providing background. It also explains the rationale behind the selection of this region along with describing the context and ethnicity in Iraq and the Kurdistan region.

Chapter two reviews relevant existing literature and elaborates on the cultural factors that impact cross-cultural negotiations.

Chapter three explains the research method and design approach used in this study. The chapter also describes the data collection and analysis process, including the sampling process.

Chapter four presents findings and discusses them in detail in light of the literature review. Six main themes emerged that show how the cultural factors impact crosscultural negotiations.

Chapter five provides the conclusions of this study. It summarises the findings of this research, discusses limitations and provides directions for future research.

#### Chapter 2

#### **Literature Review**

#### Introduction

Studies to investigate cultural factors that impact cross-cultural negotiations seem to be complex and the developments in the academic field are dynamic. As suggested by Manrai and Manrai (2012), a great variety of conceptual models and frameworks have been proposed by researchers (Brett, 2000; Foster, 1992; Ghauri & Usunier 1996; Graham, 1985; Salacuse, 1991; Sawyer & Guetzko, 1965; Usunier, 1996; Weiss, 1994) to determine the relationship between culture and international business negotiations. There is a certain density in this research area and a lack of agreement among scholars. This becomes obvious when determining what cultural factors influence cross-cultural negotiations in the current Iraqi international business environment. This research attempts to identify the main themes from cultural factors that impact cross-cultural negotiation.

A single theory or model does not seem to be fully able to cover all the cultural differences that influence cross-cultural negotiation in the current international business environment in Iraq. Consequently, this study attempts to adapt the theoretical frameworks by Salacuse, 1991; Hall, 1989 and Hofstede 2001, in order to explore the cultural differences. The focus is on the comparison of Kurdish-Iraqi and Western senior managers operating in the oil industry of northern Iraq. The researcher expects that combining the different characteristics of these frameworks will assist to uncover cultural patterns of negotiation in the current international business environment of Iraq.

The research design developed here incorporates all of what has been considered by previous studies to explain differences in culture between negotiators. It also investigates aspects that have been neglected by past studies, such as negotiation styles. Following these frameworks the research leads to six characteristics to explain cultural factors in international business negotiations: negotiation goal, negotiation attitude, communication, view of time, emotionalism and risk propensity. I define this as my research model of the cultural factors that have significant influence on negotiation behaviour. More explanations of the proposed theoretical

model are presented in the later sections of the literature review to discuss the connection between the dimensions shown in the model below and the intercultural negotiation process.

#### **Negotiation Goal**

Many studies consider negotiations a goal-directed communication process through which interdependent parties seek to achieve a mutually satisfactory outcome (Putnam & Roloff, 1992; Meina, 2011). Negotiators often enter a negotiation with motivational goals, which influence their cognitive schemas such as perception of fairness, information processing and negotiation outcomes (De Dreu & Van Lang, 1995; Van Kleef & De Dreu, 2004; Tjosvold, 1998; Carnevale & Probst, 1998). In general, the previous findings regarding negotiator motivation have noted that it is associated with cultural values but little research has assessed how culture may differ in the ways in which negotiator goals and motives influence their actions in the bargaining process. Moreover, a few studies in relation to negotiation goals have found that cultures influence on cognition and behaviour is complex and contextual (Adair et al., 2003; Brett & Okumara, 1998, Liu, 2008).

Recent research identified that negotiators need to compete and cooperate with each other to maximize their individual and joint profits and tend to pursue multiple and sometimes conflicting goals that are not only oriented towards a tangible benefit but also to attain or maintain a relationship with their counterpart (Keck & Samp, 2007; Meina & Wilson, 2011; Wilson & Putnam, 1990). Numerous studies emphasised that the relationship between negotiating parties often interacts with collectivism and individualism in activating domain specific cultural schemas to guide the negotiator's choice. Cultures that are more sensitive to relationships and subtle changes in situations can adopt different approaches when the relational contexts change (Masuda & Nisbett, 2001; Morris & Fu, 2011).

Several studies found that in conditions in which the negotiator is highly accountable for the outcome, collectivism promotes cooperative behaviour, whereas in low accountability conditions, collectivism is negatively associated with cooperative behaviour (Gelfand & Realo, 1999; Meina & Wilson, 2011). For example, Gelfand

and Realo (1999) noted that for members of collectivist cultures negotiation is a situation that involves openly confronting the other party concerning issues and is therefore inherently competitive. There is evidence that unless a norm-enhancement mechanism – such as a negotiator's accountability to others – is activated, collectivist negotiators will exhibit the collectivism-based, culturally typical cooperation (Meina & Wilson, 2011).

A few studies found that negotiators also pursue identity and relational goals that may change, impede or facilitate their instrumental goals and are found to have a significant influence on a negotiator's strategic choice as well as negotiation outcomes (Liu & Wang, 2010; Wilson & Putnam, 1990). Recent research by Meina and Wilson (2011) found these interactional goals arise from contextual features of a bargaining situation; the motivational orientation and strength of these goals are found to have a significant influence on a negotiators strategic choice as well as negotiation outcomes. In addition, goal pursuit may differ across cultures as parties with the same set of goals and motives are likely to prompt different behavioural sequences for members of different cultures (Masuda & Nisbett, 2001; Meina, 2011, Morris & Fu, 2001).

#### **Negotiation Attitude**

A large number of studies investigated the individual differences in distributive and integrative negotiation (Neale & Northcraft, 1989; Pruitt & Carnevale, 1993; Lewicki et al. 1994; Hermann & Kogan, 1977; Rubin & Brown, 1975; Terhune, 1970). Several researchers theorised that personality characteristics explain why bargainers behave the way they do during a negotiation encounter (Barry & Friedman, 1998; Rubin & Brown, 1975). The findings of Rubin and Brown (1975), suggested that negotiators enter into a relationship with the other party in which they bring variations of prior experience, background and expectations that may affect the manner and effectiveness with which a negotiation is conducted. However, some researchers consider that there is inconsistent support for the personality and negotiation relationship link (Barry & Friedman, 1998; Wall & Blum, 1991; Greenhalgh 1985; Rubin & Brown, 1975).

Previous research has suggested that the inconsistencies in the personality-negotiation research can be attributed to the negotiator's culture as culture provides negotiators with a foundation for managing social exchanges (Brett & Okumura, 1998). Early studies by Hall and Hall (1987), found that meaningful interactions only occur through the medium of culture. Studies indicate that the negotiation behaviour of individuals vary due to the underlying differences inherent in each other's culture that provide order and direction to human interaction (Clark, 1990; Kluckhohn & Strodtbeck, 1961; Limaye & Victor, 1991; McCrae, 2000).

In addition Parnell & Kedia (1996) observed that the attitudes, laws, values and traditions embedded in the cultural context of negotiators impact on the various stages of the bargaining activity. There is evidence that Hall's (1989), high/low communication context and Hofstede's (1993), individualism/ collectivism dimensions have decisive implications for the negotiation encounter (Cai et al., 2000; Morris et al., 1998; Graham et al., 1994; Mintu-Wimsatt, 2002).

For example, studies conducted by Gelfand and Christakopoulou (1999), found that individualism refers to the tendency to address self-serving concerns and individualists' elaborate self-knowledge instead of knowledge about others. In a similar vein Cai et al. (2000), reported that collectivists promote interdependence with others and focus on relational harmony and protection of group interests.

The findings of Perdue and Summers (1991), suggested that negotiators are often faced with two basic strategies: problem-solving and aggressive bargaining. Likewise, some studies have shown that problem-solving or integrative strategies yield win-win outcomes whereas distributive strategies typically generate win-lose situations (Goering, 1997; Perdue & Summers, 1991; Walton & McKersie, 1965). Due to the positive outcome generated by problem-solving bargaining numerous studies have focused on this approach and comparatively few studies have been conducted on distributive bargaining (Mintu-Wimsatt, 2002).

Similarly, the research of Goering (1997) identified that, for the integrative problem-solving, strategy communication is a critical aspect. It has been widely observed that the problem-solving approach embodies cooperative behaviours that allow negotiators to gather additional information and to correct misperceptions of the other parties' interests (Barry & Friedman, 1998; Perdue & Summers, 1991).

Similarly, it has been observed that the problem-solving approach represents certain attitudes such as the negotiator's willingness to make concessions or to adapt to the concessions of the other party (Graham et al., 1994; Westbrook, Gosh & Dev, 1996). Most of the research regarding the problem-solving approach suggests that negotiation parties cooperate with each other to secure mutually beneficial outcomes by identification of common goals (Perdue & Summers, 1991; Perdue, 1992; Walton & McKersie, 1965; Goering, 1997). Evidence suggests that the problem-solving approach is beneficial to establish and maintain successful long-term relationships.

#### Communication

Over the past decade numerous studies examined the meaning and context of negotiations and how culture shapes the substance of communication (Adair et al., 2001; Brett & Okumura, 1998; Cohen, 1997; Fisher et al., 1991; Kopelman & Olekalns, 1999; Leung, 1997; Markus & Kitayama, 1991). In general, the findings of previous work often defined negotiations as a back and forth communication process with the aim or reaching mutually beneficial outcomes (Adair et al., 2001; Brett & Okumura, 1998; Kopelman & Olekalns, 1999).

Many studies supported the idea that differences in communication styles are often considered as barriers towards integrative processes and negotiation outcomes (Buttery & Leung, 1998; Drake, 2001; Morris et al., 1998; Putnam & Jones, 1982). In addition, the research of Putnam and Jones (1982), showed that communication can serve both integrative and distributive functions in the negotiation process. Some studies reported that, of the many dimensions of cultural variability, the most relevant findings of the negotiation literature have been individualism-collectivism, power distance, communication context and conception of time (Brett & Okumura, 1998; Cohen, 1997; Leung, 1997). Some researchers found that communication is not only concerned with how interaction is shaped by cultural values but also by contextual factors (Hunter, 1998; Kim, Shin & Cai, 1998).

The early findings of Hall (1989) on the cultural context of communication pioneered a vast array of research on the impact of context on the negotiation encounter. For example Hall (1989), discovered in his studies that low

communication cultures use explicit, direct language compared to high communication context cultures that use explicit indirect language. The research of Simintrais and Thomas (1998) indicated that low-context cultures rely on formal communication that is often expressed verbally with the informal context being less important in understanding the message. Similarly, studies conducted by Keegan (1989), suggested that in high-context cultures less information is contained in verbal expression and therefore context variables such as values and position in society need to be considered to comprehend the message. As a result the message is bound to the implicit context in which it is delivered (Keegan, 1989).

In addition, a number of researchers reported that under conflict conditions, low-context communication negotiators are likely to rely more on direct verbal strategies while high-context negotiators may employ more indirect communication approaches including third party intervention for tension reduction (Augsburger, 1992; Cohen, 1997). Moreover, Ting-Toomey and Kurogi (1988) suggested that low-context conflict communication aims clearly to state the speaker's true intentions while high-context communication focuses on concealing the speaker's true intentions.

In a similar vein, communication research supports the idea that individualistic cultures tend to stress the value of directness in their communication approach and verbalise their individual wants and needs, while members of collectivist cultures tend to stress the value of contemplative talk and discretion in voicing their opinions and feelings (Gudykunst et al., 1996; Kim 1994; Trubisky et al., 1991). Likewise, the findings of Trubisky et al. (1991) suggested that negotiators from individualistic cultures tend to prefer direct conflict communication and solution-oriented styles, emphasising the values of autonomy, competiveness and the need of control. Whereas representatives of collectivist cultures tend to prefer obliging and conflict avoidance styles that emphasise the value for maintaining relational harmony in conflict interactions.

In general, research on negotiation and communication supports the idea that communication has a positive impact on integrative negotiation as judgment errors are corrected through information exchange aimed at exposing the mutual compatibilities of the parties (Cross, 1977; Pruitt & Lewis, 1975; Tutzauer & Roloff,

1988). For example, Thompson (1990) found that negotiators who sought information about others' priorities achieved higher profits. According to their results, information-seeking was positively reciprocated and, even when only one side shared information, joint profits improved.

#### View of Time

A large number of studies reported that the concept of time and the value of time vary across different cultures and can have a considerable impact on cross-cultural negotiations (Bhagat & Leonard, 2002; Bluedorn, Kaufman & Lane, 1992; Foster, 1992; Hall & Hall, 1987; McDevitt, 2006; Ting-Toomey, 1999). In general, previous findings regarding the cultural view of time found that time has a direct effect on the negotiation process. The reason for this is that some cultures consider time as a limited resource and therefore a valuable commodity that should be used wisely. Specified time frames are important to these cultures. By contrast other cultures, such as Kuwait, Egypt etc. view time as boundless and plentiful and their approach to the negotiation process may be somewhat slow. For example, Hofstede (1993) found that American negotiators view time as a scarce commodity and thus it should be used with maximum efficiency. In comparison Japanese and other Asian counterparts consider time in terms of the long-term perspective.

The early findings of Hall and Hall (1987), suggested that polychronic cultures take longer to make decisions. They do not discriminate among issues or does they prioritise them according to a Western logical flow. McDevitt (2006) observed in her research that in polychronic cultures all matters receive full attention at the time they come up, which can cause a delay in the negotiation process. According to Hall and Hall (1987), monochronic cultures perceive and use time in a linear way, allowing a person to concentrate on one issue at a time. In addition, some researchers found individualistic cultures tend to be associated with monochronic time, whereas collective cultures embrace polychronic time (Hofstede and Usunier, 1996).

Numerous studies reported that a culture's attitude towards time determines the importance placed on the development of interpersonal relationships in business (Adler, 1986; Boone & Kurtz, 1996; Bovee & Thill, 1995; Foster, 1992; Hall, 1989).

For example, Ting-Toomey (1999), reported that in polychronic cultures more emphasis is placed on forming relationships rather than holding to a tight schedule. The reason for this is that in polychronic cultures time is experienced as much less tangible as in monochronic cultures (Ting-Toomey, 1999). By contrast, some researchers found that monochronic cultures with their time consciousness give less priority to building long-term relationships.

It has been widely observed that monochronic cultures are concerned with causality and argument and communication are based on the need for logic (Adler, 2002; Boone & Kurtz, 1996; Bovee & Thill, 1995; Foster, 1992). In addition, numerous studies reported that polychromic societies are concerned with equilibrium and argument and general communication are based on the need for balance, emphasising the here and now (Boone & Kurtz, 1996; Bovee & Thill, 1995; Gudykunst & Kim, 1992; Harcourt, 1996; Ober 1995). In addition, the above studies found that individuals from monochronic cultures tend to focus on agenda setting, objective criteria and deadlines to accomplish goals. By contrast, several studies found that polychromic cultures focus their attention on improving the relational and contextual elements that frame the negotiation (Boone & Kurtz, 1996; Bovee & Thill, 1995; Gudykunst & Kim, 1997; Ober 1995).

Studies conducted by Mayfield et al., (1997) found that the preference of monochromic cultures for linearity and logic appears to be one-dimensional and sterile to polychromic cultures. By contrast monochronic cultures have been reported to find polychromic cultures illogical and unproductive.

#### **Emotionalism**

Many research studies assessed the role emotions play in negotiation. They report that due to the influence on cognition and communication it has a practical impact on negotiators' behaviour, the negotiation process and outcomes (Allred et al., 1997; Allred, 2000; Butt et al., 2005; Davidson & Greenhalgh, 1999; Kopelman et al., 2006; Morris & Keltner, 2000; Thompson, Nadler & Kim, 1999). A large number of studies recorded the beneficial and unfavourable effect of emotions on joint outcomes but few empirical studies assessed how culture interacts with emotions in

shaping individual behaviour choices in negotiation situations (Carnevale & Isen, 1986; Van Kleef & De Dreu, 2002; De Dreu, Manstead & Van Kleef, 2004; Liu, 2008). However, it has been widely observed that the interpersonal effects of emotions can be used to strategically influence the other party in order to achieve substantive or relational goals in negotiation (Adler et al., 2002; Kopelman et al., 2006).

Early studies identified that the conflict and bargaining that occurs during negotiations has an inherent affective component and emotional reactions such as distress or anger are often experienced during the negotiation process (Purdy, 1967; Walton & McKersie, 1965). The research of Adler (1986), suggested that the emergence of negative spirals in cross-cultural negotiations that cause increasingly negative emotional reactions in negotiators will hurt the negotiation process and often bring it to an end. For example Kumar (1979), found in his study observing negotiations between Japanese and US negotiators that negative feelings in negotiations cause less cooperative and more hostile interactions. This makes it less likely to reach integrative bargaining solutions. Conversely, Van Kleef et al., (2002) found through computer-mediated tests that negotiators with an angry counterpart make more concessions than did those with a non-emotional counterpart.

The research of Mesquita and Frijida (1992) on the relationship of culture and emotion identified important differences in the way emotion is expressed across cultures. According to their results cultures differ in the degree to which emotional expressiveness and emotional restraints are valued and practiced in social interactions around negotiations. For example, Trompenaars and Hampden-Turner (1998) reported that members of cultures that are emotionally expressive tend to visibly demonstrate their feelings through body posture and facial expressions. They suggested that emotionally expressive cultures tend to value affective engagement and involvement in communicating with others. By contrast, Ting-Toomey (1999), reported in her work that members of cultural systems valuing emotional restraint tend to contain, hide, mask or minimise more overt emotional expressions.

Studies have shown that negotiators respond to expressed anger by their counterparts in either displaying similar behaviour or demonstrating complimentary behaviour that is viewed necessary to maintain the interaction (Butt et al., 2005). According to

recent research by Liu (2008) anger may convey the impression that a negotiator will not settle for a suboptimal outcome by pressing the counterpart to be less committed to original positions but also the counterpart will lose motivation to explore creative options for mutual gains. Research by Allred et al., (1997) suggested that when a bargaining situation has potential for "win-win" outcomes, tendencies suggest a less productive bargaining dynamic with smaller joint gains and less desire to work in the future. Whereas some studies reported that in highly competitive situations it may increase the chance of premature impasses and allow the angry negotiator to gain the upper hand (Pillutla & Muriningham, 1996; Sinaceur & Tiedens, 2006).

However, the research by Lewicki et al., (2010) supported the idea that negotiators either adopt a distributive approach and focus on claiming more value for themselves or use an integrative approach to create more value for both parties. There is evidence that distributive strategies – including threats, positional commitments and persuasive arguments – tend to yield win-lose outcomes and smaller joint profits (Pruitt & Lewis, 1975). In addition, Thompson (1990) found that integrative strategies such as information exchange, multiple-item offers and relationship-building tend to yield win-win outcomes, higher joint profits and greater desire to cooperate in the future.

#### **Risk Propensity**

Numerous studies reported that risk and uncertainty are generally present in the context of negotiations and differences and the perception of risk has direct implications for the exchange of negotiators (Neal & Bazerman, 1992; Deutsch, 1973; Warner, 1995; Weber, 1994; Markowitz, 1959; Weber & Hsee, 1998; Ordonez et al., 2009). In general, previous research supports the idea that uncertainty avoidance describes a culture's attitude towards risk and uncertainty regardless of whether it embraces or attempts to develop mechanisms to control and limit the uncertainty (Douglas & Wildavsky, 1983; Hofstede, 1980; Hofstede & Bond, 1984; Hofstede, 1993; Hofstede, 2001; Bond & Luk, 1993).

The findings of Hofstede (1993) suggested that in high-uncertainty avoidance cultures individuals are motivated by security and desire to control and predict the

future whereas in low-uncertainty avoidance cultures individuals accept the future. They are motivated by success and are more willing to take risks. Douglas and Wildavsky (1983) found that each society has a general preference for certainty versus uncertainty and risk avoidance versus risk seeking. In addition, Douglas and Wildavsky (1983), defined risk as a set of social processes that is understood though cultural analysis and interpretation. The reason for this is that each type of risk is exaggerated or minimised, feared or embraced according to the social and cultural acceptability of the underlying activities.

Early research suggested that certain organisational cultures are found to systematically encourage or discourage risky behaviour of individual members or developed institutionalised responses to risk (Deal & Kennedy, 1982; Douglas & Wildavsky, 1983; Heath et al., 1999). In similar vein, March and Shapira (1992) identified that an organisation's control systems serve to influence the risk behaviour of individual members and reward the outcomes achieved and the processes used in the risky decision-making.

Likewise, some studies at group level establish substantial congruence between a society's value orientation towards risk and the nature and the degree of decision shifts enacted by groups (Carlson & Davis, 1971; Neale & Bazerman, 1992; Stoner, 1968). Generally the above studies reported that cultures that value risk-taking will become more risky in their group decisions and cultures that value risk aversion will become more careful in their group decisions. For example, Teger and Pruitt (1967) observed that the American culture values risk and Americans believe themselves to be risk-takers but have observed that other cultures have similar levels of risk-taking.

Some researchers found that an individual's cultural orientation towards handling risks in specific contexts tends to persist over time, forming a relatively stable pattern (Brockhaus, 1980; Wallach & Kogan, 1965; Slovic, 1964; Sitkin & Pablo, 1992). Conversely, Slovic (1964) reported in his research that risk orientation is not a unified characteristic or trait. For example, Goering (1997) discovered that groups within the same culture were shown to shift in both directions depending on the specific situation or the specific problem.

A few studies found that propensity towards risk is an individual variable that has been found to affect negotiation processes and outcomes (Locke & Latham, 2002;

Neale & Bazerman, 1985; Sherman & Kiresuk, 1968; Shure & Meeker, 1967; Thompson, 1990). For example, Farber and Katz (1979) reported that goals reflecting a risk adverse orientation were found to increase concessionary behaviour and increase joint gains in negotiations, while goals reflecting risk-seeking orientation increased contentious behaviour. In addition, some studies found that negotiators that consider negotiations as a win-win process were found to be more risk averse, while win-lose negotiators were found to be more risk seeking (Neale & Bazerman, 1985, Ordonez et al., 2009).

#### **Summary**

Research into cross-cultural negotiation suggests a great variety of conceptual models and frameworks which have helped to determine the relationship between culture and international business negotiations to this date. The complexity of this area and the lack of agreement among scholars are obvious when asking what are the cultural factors that influence cross-cultural negotiations in the current international business environment in Iraq. Despite the lack of agreement in the research, certain key cultural themes have emerged which have been described in this chapter.

In this chapter, the researcher followed the suggestions and theoretical frameworks by Hall, (1989), Hofstede (2001) and Salacuse (1991) in order to structure the literature review and look at cultural factors that influence cross-cultural negotiations. Following these theoretical frameworks, six main factors that impact cross-cultural negotiations were identified: negotiation goal, negotiation attitude, communication, view of time, emotionalism and risk propensity. A close look shows that the majority of these concepts are inter-linked.

For this reason the participants in this study were asked to reveal their attitude towards these cultural factors. It was hoped that the application and adaption of the three frameworks would lead to comprehensive and reliable findings concerning the cultural factors that influence cross-cultural negotiations in this environment. The next chapter will explain how the research project is designed and how the data was collected

#### Chapter 3

#### **Research Design**

This chapter on the research design outlines for the reader the value and validity of the research and its contribution to the body of knowledge (Creswell, 2002). Sekaran (2003) defined research design as an academically established regulatory framework with the intention of collecting and evaluating existing knowledge in order to arrive at, and validate, new knowledge. Mishler (1990) suggested (as cited in Miles & Huberman, 1994, p. 5), that "No study conforms exactly to a standard methodology; each one calls for the researcher to bend the methodology to the peculiarities of the setting." According to Cooper and Schindler (1998), the quality and value of research is determined by the extent to which the methodology is articulated, as well as the selection of the most appropriate research approach. Thus, this chapter outlines the research design and justifies the methodology adopted to collect and process the data for this study.

This chapter also introduces and discusses the methodological approach adopted for this study as well as the analytical approach used for the data collection and analysis of the material.

#### Research aim

Research scholars have identified exploratory, descriptive, explanatory and prescriptive purposes for the research activity (Saunders, Lewis & Thornhill, 2000). This current research adopts an exploratory approach and aims to match the research questions to the research objectives.

According to Saunders et al. (2000), exploratory research can be conducted through structured and semi-structured interviews with expert informants. The design of the research and interview questions is based on a review of the relevant literature. The main purpose of exploratory research is the exploration of a complex research problem or phenomenon with the intention to clarify the identified problems. It can also investigate the nature of a selected phenomenon while uncovering new knowledge (Robson, 2002). The research intention is explorative: A research

question has been formulated: "How are cross-cultural negotiations managed in the current international business environment of northern Iraq.?" In order to answer this question it was necessary to explore the variables that normally influence cross-cultural negotiations. Thus, the review of the relevant literature has formed the basis of this project. The analysis of the data that was collected will be a primary focus of this project.

According to Creswell (2003) the selection of the research approach is an essential decision that influences the quality of the study. It gives the researcher the opportunity to consider different approaches to satisfy the objectives and limit the study. The approach chosen in this research was to target individuals from local area with Iraqi and Western cultural backgrounds. These individuals are operating in senior management positions in the oil industry of northern Iraq or for the Kurdish Ministry of Natural Resources (KMNR). Participants were selected from a KMNRissued list of companies, providing oil and gas related services in the Kurdistan region. Participants were divided into two groups, a local Kurdish-Iraqi and an aggregate Western group. All candidates had been working in the region between 7 and 37 years. All participants had experience with the local culture and local negotiation practices. Eight in-depth interviews with members of this target group were conducted. These interviews provide sufficient data to identify themes of concern. The themes showed commonalities with the themes from the intercultural negotiation literature. The researcher identified that there is a knowledge gap concerning research on the impact of culture on cross-cultural negotiations. This is particularly so concerning the international business environment of northern Iraq and little has been published to date. Therefore, the findings of this study are interesting and will contribute new knowledge and to the literature and theoretical developments on intercultural negotiation.

#### **Philosophical Approach**

The philosophical assumptions that a researcher adopts shapes the use of abstract ideas and beliefs that inform the research (Creswell, 2013). As suggested by Myers, (2009), the most pertinent philosophical assumptions are those that relate to the underlying epistemology or ontology that guides the research.

Ontology relates to the nature of reality and its characteristics, when conducting qualitative research, the researcher embraces the idea of different realities. Epistemology or the "theory of knowledge" attempts to answer the question what counts as knowledge and how knowledge claims are justified (Creswell, 2013). A researcher with an epistemological assumption conducting qualitative research tries to get as close as possible to the participants being studied to minimize the distance between himself and those being researched (Creswell, 2009). Denzin and Lincoln (2011), consider the philosophical assumptions of ontology and epistemology as key premises that are folded into interpretive frameworks. The literature broadly classifies three paradigms for qualitative research: positivism, constructivism and critical theory.

Positivism is the dominant form in most business and management disciplines. Positivist researchers generally assume that reality is objectively given and can be described by measureable properties that are independent of the researcher (Bryman & Bell, 2011). As suggested by Myers (2009), the role of positivist research is to test materials for the development of laws to increase the predictive understanding of phenomena. Myers (2009), noted that positivist researchers usually formulate propositions that portray the subject matter in terms of independent and dependent variables and the relationship between them.

Interpretivists hold the view that it is important to explore the subjective meanings that motivate people's actions in order to understand their socially constructed reality. This cannot be understood independent of the actors who make that reality (Saunders et al., 2000). As a result, interpretivist researchers normally employ qualitative research techniques such as semi-structured interviews with the intention not only to reveal and understand the "what" and "how" but more importantly to place emphasis on exploring the "why" (Saunders et al., 2009). In other words, an exploratory approach allows the researcher to make sense of a social phenomenon. This is not only through interaction with their environments but also by making sense of it through the interpretation of the events and meanings they draw form it (Saunders et al., 2009). Similarly, an interpretivist approach allows the researcher to overcome the barriers that separate him- or herself from the participants. For this reason interpretive researchers do not define dependent and independent variables

but focus instead on human sense making to understand phenomena through the meaning people assign to them (Kaplan & Maxwell, 1994).

Critical theory is a philosophy that is based on the interdependency of human values and beliefs (Johnson and Christensen 2010). Critical researchers assume that social reality is historically embedded and that it is produced and reproduced by people. As suggested by Myers (2009), people can consciously act to change their social and economic circumstances but their ability is constrained by various forms of social, cultural and political domination. The main task of critical research is to use social critique with the intention of bringing restricting and alienating social conditions to light. Instead of describing current knowledge and belief the intention is to challenge prevailing belief, values and assumptions (Myers, 2009).

This research takes an epistemological interpretivist stance in order to explore how successful cross-cultural negotiations are conducted in the international business environment of northern Iraq. The researcher believes that reality is only obtained through relationship with the research subject by employing qualitative research techniques. An interpretive approach enables the researcher to reduce barriers between himself and the participants in order to obtain appropriate knowledge (Bryman & Bell, 2011). Furthermore, this research has taken an ontological interpretive position regarding the participants' perceptions of cross-cultural negotiation. This research embraces the premises that a social world exists which is constructed and shaped by the experiences of the participants (Limpanitgul et al., 2009).

To sum up, this research has taken an epistemological and ontological interpretivist stance in order to attempt to explore and interpret the social interactions between Kurdish-Iraqi and Western senior managers.

#### **Data Collection**

This research utilised primary data from semi-structured, in-depth interviews and secondary data from official documents and reports, published scholarly articles. The following section addresses the collection of primary data.

Semi-structured, in-depth interviews were used as a means of collecting primary data in this study. Myers (2009) suggests that interviews are one of the most important

and most useful data gathering techniques in qualitative research. Interviews allow the researcher to collect rich data from people in various roles and situations in order to see what is not ordinarily on view and to examine its deeper meanings (Robson, 2002). Face-to-face interviews enable researchers to establish a strong rapport and to interact and gain more information from the participants (Myers, 2009). In addition, as suggested by Angrosino (2007) (as cited in Creswell, 2013, p. 166), "Observation is one of the key tools for collecting data in qualitative research. It is the act of noting a phenomenon in the field setting through the five senses of the observer, often with an instrument, and recording it for scientific purposes." Observations are based on the research purpose and question. The researcher observed the participants during the interviews. Observing the participants during the face-to-face interviews allowed the researcher to arrive at a more complete picture of the phenomenon and its environment, in order to more effectively evaluate the interview data (Myers, 2009).

#### The interview questions

A set of questions was used in this study. The interview questions aimed to gather useful and relevant information to answer the research question. The research interviews were semi-structured and, as suggested by Myers (2009), involved the use of some pre-formulated questions, but there was no strict adherence to them. Table 1 shows the indicative interview questions adapted from Salacuse (1991), which encouraged participants to share their experience and expertise of the cross-cultural negotiation process.

The interviews were based mainly on the research questions along with some exploratory questions when the initial response from the participant was too short. For example, "Would you please elaborate on the topic?"

#### Table 1

#### Interview Questions

#### Define the research question

To explore the view of participants about cross-cultural negotiations in the current international business environment of northern Iraq.

- 1. Can you please tell me how you conduct a negotiation?
- 2. Can you please tell me, when negotiating, what are your goals?
- 3. Can you please tell me, what is your attitude towards negotiations?
- 4. Can you please tell me, during your negotiations, how do you communicate with your opposite?
- 5. Can you please tell me what value has time for you when negotiating?
- 6. Can you please tell me, when negotiating what is your opinion of emotions at the negotiation table?
- 7. Can you please tell me, how would you describe the nature of your agreements?
- 8. How do you build up your agreements? What are the steps involved?
- 9. Can you please tell me, do you involve other colleagues or employees in the decision making process?
- 10. Can you please tell me, do you feel comfortable to take risks?
- 11. Can you please tell me, when negotiating; is protocol or formality important for you?

Note. Adapted from Salacuse (1991)

#### **Interview process**

As described by Kahn and Cannell (1957), an interview is a purposeful discussion between two or more people and is a useful tool to gather valid and reliable data. Morse (2011) suggested in order to collect sufficient in-depth data for qualitative research a minimum number of six interviewees is needed. For this reason, interviews were conducted with eight participants in semi-public places such a government offices, company headquarters and coffee houses. The interviews were conducted following this particular approach. At first, potential candidates were contacted either via email or by phone depending on the available contact details. After the initial contact with the participant was established they were emailed and provided with the Participant Information sheet (see Appendix A) and a digital copy of the consent form (see Appendix B).

The time and location of the meetings were decided to suit the convenience of the participants in order to fit into their schedules. At the beginning of each interview the researcher explained in detail the participant's rights and asked each one to familiarise themselves with and sign the consent form.

Each interviewee was asked about how he or she reacts in particular conflict and negotiation situations, as well as the importance of particular cultural factors with regards to any negotiation process the participant was confronted with within the last 12 months. The interview consisted of twenty-five questions. Individual answers were measured against a total score based on an interpersonal conflict questionnaire (see Appendix C) with similar questions from the area of applied psychology (Rahim & Magner, 1995). Next, participants were asked to tell the interviewer about an important negotiation process they were confronted with within the last 12 months. This was followed by 11 open-ended questions (see Appendix D) that were adapted from the work of Salacuse (1991). The questions were grouped into the categories of conflict (competing, accommodating, avoiding, collaborating, style compromising) and the cultural factors influencing negotiation (Rahim & Magner, 1995; Salacuse, 1991). Questions on the category of conflict style were more explicit while questions on cultural factors were more general in order to gain a deeper knowledge on the topic as well as to have the freedom to ask questions that were not mentioned in the interview guide without losing rapport. This approach allowed the participants to give feedback about the way in which information would be used. On average the duration of the interviews were about one hour and ranged from 20 minutes up to 90 minutes. The scope of the answer was left to the discretion of the participants, however, the participants were encouraged to elaborate and provide more information.

#### Secondary data

Secondary data sources included official documents and reports, available from the International Energy Agency (IEA), Kurdish Ministry of Natural Resources (KMNR) and publicly available books on the history of the region. For the relevant scholarly articles required for the literature review the researcher utilised online databases and local libraries.

#### **Participant sampling**

In this research, the participants were recruited via a strategy of purposeful sampling in order to approach Western and Kurdish-Iraqi senior managers operating in northern Iraq (Myers, 2009). The researcher, started the data sampling by contacting participants based on criterion sampling (Saunders et al., 2000).

This method was appropriate in order to identify individuals who have experience with cross-cultural negotiation in the region due to the involvement of their organisation and the position they held. In the criterion sampling, the following aspects were taken into account:

- Number in sample: Eight
- Gender: Both male and female participants (Note from prior experience working in the area, I was aware that female managers were working in the region.)
- Nationality: Western or Kurdish-Iraqi
- Involvement of their organisation in cross-cultural negotiation
- Job position: Senior manager
- Length of stay in the region: Minimum of four years
- Area: Northern Iraq (Erbil, Sulaymaniyah, Dohuk)

Criterion sampling is particularly useful because all the individuals studied represent people who have experienced the phenomenon of cross-cultural negotiation in the region.

## Sample size and demographics

For this research, eight participants were identified through their business presence in northern Iraq. Representatives of the oil industry firms, government agencies and industry bodies were selected due to their expertise; meaning, that is where they have specific knowledge or opinions of this area. All representatives were approached by the researcher based on a publicly available list of suppliers of oil and gas-related services that have been recommended by the domestic Kurdish Ministry of Natural Resources (KMNR). All eight participants had different cultural backgrounds, being American, Austrian, German or Kurdish but were asked to describe their orientation, when asked, as either Western or Kurdish-Iraqi. Their ages ranged from early thirties to mid-fifties with the majority being aged between thirty and forty years. The participants' working experience in the region ranged between four and thirty-seven years. All but one of the participants were male managers due to the Islamic background of the areas; women in senior management positions are only represented in small numbers. Male and female senior managers in this study exhibited similar values and traits. A plausible explanation for this are the findings originating with Ashforth and Mael (1989) on social identity that suggest that female managers in male-dominate organisations adopt behaviours, opinions, attitudes and beliefs typical of male managers in order to minimise differences in perceptions.

At first contact, an email with the attached interview questions was sent out to notify potential candidates of the interview content. The intention was to give potential candidates the opportunity to familiarise themselves with the structure and questions of the interview in order to make an informed decision whether or not they were willing to participate. In addition to these eight interviews in the Kurdistan oil and gas service industry, Petroleum Geologist and adviser to the Kurdish Minister of natural resources, Mr Atruschl Mohamed was contacted with regards to background information on the northern Iraqi oil industry. He and his office were willing to offer their experience and give recommendations on the topic.

The following table lists the details of the participants:

Table 2

Details of the participants

Organisation / Government Body	Number of Participants	Number of Interviews	Organisational Role of Participant	Approximate Time in North Iraq	Ethnic Background	Gender
International Oil Company	1:	1	Drilling and Operations Manager	4 1/2 years	Western	Male
International Oil Company	1	1	Country Manager	4 years	Western	Male
International Oil Company	1	1	Country Manager	6 years	Western	Female
International Oil Company	1	1	Chief Financial Officer	5 years	Western	Male
Domestic Oil Company	1-	1	Owner / Principal Partner	Born in Iraq	Kurdish/Iraqi	Male
Government Body	1	1	Project Leader	Born in Iraq	Kurdish/Iraqi	Male
Government Body	1:	1	Associate Director	Born in Iraq	Kurdish/Iraqi	Male
Domestic Oil Company	1_	2	Commercial Manager	Born in Iraq	Kurdish/Iraqi	Male

Details of Participants

## **Analytic Approach**

For analysing the interviews grounded theory offered the most flexibility as the research was exploratory (Glaser, 1978). As suggested by Saunders et al. (2009) exploratory research unfolds through focus group interviews, structured or semi-structured interviews with experts and a search of the relevant literature. Thereby exploratory research investigates a specified phenomenon for the purpose of exposition and uncovering new knowledge (Robson, 2002). Grounded theory is useful for analysing data in exploratory studies as it relies on the production of theoretical perspectives derived from data. As suggested by Glaser (2005) grounded theory is a tool in order to discover what is going on within a particular arena. It is a particularly valuable method for studying social and psychological phenomena or other areas about which little is known (Wuest, 2007). Furthermore, the aim of grounded theory is to generate theory that accounts for patterns of behaviour that is important and problematic for all parties involved (Glaser, 1978). It follows that for

this reason, and because "grounded theory captures social process in social context" (Wuest, 2007, p. 244), this method is useful for explaining the impact of culture on intercultural negotiation.

The grounded theory framework is very beneficial in developing context-based, process-oriented descriptions and explanations of social and organisational phenomena (Myers, 1997). According to Urquhart (2001), it offers well-signposted procedures for data analysis and allows for the emergence of original and rich findings that are closely tied to the data. The researcher adopted the grounded theory analysis framework in order to analyse the data, as described by Myers (2001), and as shown in Table 3.

Table 3

Phases of grounded theory

Steps of Grounded Theory	Outline of the process	
1. Data Collection	Researcher has to collect qualitative data and transcribe it.	
2. Open coding	Researcher analyses a sentence or paragraph of the text and summarises this text by the use of a succinct code. Open codes are descriptive, analysis has not proceeded beyond identifying concepts at the level of 'categories'. Constant comparison is most important activity.	
3. Selective Coding	Researcher refines the conceptual constructs to explain the interactions that occur between the descriptive categories.	
4. Theoretical Coding	Researcher aims to create inferential and/or predictive statements about the phenomena. This is achieved by specifying explicit causal / correlational linkages between individual interpretive constructs.	

Being an inductive approach, qualitative data analysis begins with a complete set of collected data in the form of text (Wuest, 2007). In this case, the researcher transcribed the tapes for all the interviews. In the initial stage of the analysis the researcher read through all the data collected in order to stimulate theoretical thinking and analytical strategies. The first step of the data analysis was coding the interview transcripts and this helped the researcher to move away from particular statements to more abstract interpretations of the interview data (Charmaz, 2006). After coding several interview transcripts the researcher identified issues that were important to the respondents and assigned them a conceptual label known as

phenomena. As some codes shared similar characteristics they were pooled together in categories, which were interlinked and used as the basis for an emerging theory. The emerging theory was then compared with the extant literature, in order to improve construct definitions, similar frameworks with the intention to improve external validity. As Dick (2002) suggested, in this way no theory is forced on the data but the data provides the basis for new emerging theories.

#### Limitations

The research was restricted to Erbil, the main district of Kurdistan which is a unique region of Iraq. This is due to its Kurdish culture which does not exist in the other parts of the country. One of the biggest limitations this research faced was the limited number of interviews with participants. Also, the participants were drawn from only one industry only; other industries were not considered. Therefore the research results might be limited to only the northern Iraqi oil and oil services industry and cannot be generalised to the whole of Iraq. While this study is limited by its small sample size, this, however, is a common problem in intercultural negotiation research (Cai & Drake, 1998). Finally, a potential problem which has to be acknowledged is that the interviews were conducted in English which is a second language for many participants which might have limited the range and depth of the responses.

### **Ethical Considerations**

At the beginning of the data collection the researcher submitted an ethics application to the Auckland University of Technology (AUT) Ethics Committee with the intention of establishing the safety of the participants in the study. After approval (Ethics Application Number 13/94 dated 24 May 2013, see Appendix E) government institutions and international oil companies were contacted and informed about the scope of the research with the Participant Information sheet and a digital copy of the consent form. Before the commencement of each interview participants were made aware of their rights and that in case any confidential information was asked for they had the right to either stop the interview or ask the

researcher to keep the information confidential. Written consent from each was obtained and participation was voluntary and participants were assured of the utmost confidentiality since the researcher was not planning to ask any questions that were of commercially sensitive and that could disadvantage the participants or their companies. After the interview process and the research had been completed participants were offered a copy of the research findings.

## **Summary**

This chapter explains the methodological framework for this research. Utilising an interpretive exploratory method, the research adopted a qualitative approach to collect the data and answer the research question.

Eight semi-structured interviews were conducted with eight participants. The participants were recruited using a criterion sampling approach. Grounded theory was used to analyse the transcribed information from the interviews. Chapter four presents the data and the findings obtained through the methods and techniques described in the chapter and further discuss them in comparison to previous literature.

## Chapter 4

### **Results and Discussion**

This chapter presents analysis of the interviews and the results. This section starts with detailed results of the conflict approach of the negotiation process of Western and Kurdish / Iraqi negotiators. This is followed by the in the data identified themes and subthemes that emerged and are presented under the topic of the interview questions. Some illustrative materials from the interviews are added in italic.

Table 4 shows the key themes that emerged regarding the perceptions of Kurdish-Iraqi and Western respondents in their approach to negotiation. In order to capture the essence of what the participants reported in the research, the information collected is arranged under the following themes:

- "Goal" describes what the participants perceive the very purpose of a business negotiation to be.
- "Attitude" explains the differences in the approaches of the participants to deal-making.
- "Communication" refers to the methods of communication among the cultures of the participants.
- "Time" refers to the attitude of a participant's culture towards time.
- "Emotion" accounts of the negotiating behaviour of the two groups of participants.
- "Risk" describes the participants' attitudes towards risk-taking.

Table 4

Themes identified through analysis of the research data

Main Themes	Sub-Themes
1. Goal of business negotiations	Win-win or win-lose outcome
2. Attitude towards negotiation	Economic outcome or relationship.
3. Methods of communication	Direct or indirect methods of communication
4. Sensitivity to time	Monochronic or polychronic
5. Emotionalism	Expressive or restrained
6. Willingness to take risks	Risk-taking or risk-adverse behaviour

The following section discusses the first theme Goal This theme is presented structured into the sub-theme 'Win-win or win-lose outcome' and is presented under the research question:

# Theme 1. Goal of Business Negotiations: When Negotiating What Are Your Goals?

This question was used as an introduction to the topic and to help to build rapport with the participants. The formulation of the question is open and the respondents did not seem to have any difficulties with their replies. The intent of this question was to explore what the participants perceived as their goal when negotiating. The response was left to the participants own judgement.

Many participants spoke here about the importance of having a good working relationship with the other party, setting yourself goals you want to achieve,

ascertaining the comfort level of the other party, achieving a good economic outcome without hurting the other party in the process, and making both parties as comfortable as possible. Western and Kurdish negotiators stressed the importance of setting themselves objectives they wanted to achieve. This finding was in line with a large number of studies that suggest that many negotiators enter negotiations with motivational goals which influence their cognitive schemas such as perception of fairness, information processing and negotiation outcomes (De Dreu & Van Lang, 1995; Van Kleef & De Dreu, 2004; Tjosvold, 1998; Carnevale & Probst, 1998). For the majority of Western negotiators the objectives were to achieve the best economic outcome without damaging the relationship with the other party. This finding was in unison with recent research that proposes that negotiators need to compete and cooperate with each other to maximise their individual and joint profits and tend to pursue multiple and sometimes conflicting goals that are not only oriented towards a tangible benefit but also to attain or maintain a relationship with their counterpart (Keck & Samp, 2007; Meina & Wilson, 2011; Wilson & Putnam, 1990).

Example of replies from Western negotiators include.

Before a negotiation of course you have to set your own objectives. My primary goal is to achieve the goals I have set and to then find a good comfort level for all parties involved. (Participant B, male, Western senior manager)

I usually set myself a certain point I would like to achieve and then I try my best to achieve it. I also try to put emphasis on forming a relationship with the other party but sometimes it is difficult as many people still have this culture of bazaar heckling, even the ministries and officials. Meaning to say that they always offer you the lowest price, they are always arguing that the service you provide is not good enough. After a while they will slowly, slowly try to reach an agreement with you. (Participant B, male, Western senior manager)

In contrast, Kurdish-Iraqi negotiators put their main emphasis on forming a good relationship with the other party and make sure to achieve a good comfort level for

all parties involved at all times. This finding is consistent with several studies that emphasised that the relationship between negotiating parties often interacts with collectivism and individualism and activates domain-specific cultural schemas to guide the negotiator's choice. Moreover, it is suggested that collectivism is associated with cooperative behaviour and is more sensitive to relationships and subtle changes in situations (Masuda & Nisbett, 2001; Morris & Fu, 2001).

In negotiations you sometimes fail but the goal is to be ending the negotiation in a pleasant way for both parties, to make both happy. (Participant E, male, Kurdish-Iraqi senior manager)

My first goal is to have a good relationship and second to get as much benefit as possible for my side. (Participant H, male, Kurdish-Iraqi senior manager)

A good working relationship is important to make the other party to fulfil your demands and to cooperate. (Participant F, male, Kurdish-Iraqi senior manager)

The findings imply that both Western and Kurdish-Iraqi negotiators enter negotiations with motivational goals that influence their cognitive schemas. The relationship between the two negotiating parties interacts with collectivism and individualism as a cultural schema to guide the negotiator's choice. Therefore, as Kurdish-Iraqi participants are more sensitive towards maintaining good relationships with the other party the behaviour can be associated with collectivism as a domain specific cultural schema to guide their choice. In contrast Western negotiators are more oriented towards a tangible benefit and are less concerned with the relationship which can be attributed to individualism. This is in line with the negotiation literature that suggests negotiation is associated with cultural values, and varies according to the ways in which a negotiator's goals and motives influence his or her actions in the bargaining process (Keck & Samp, 2007; Meina & Wilson, 2011; Wilson & Putnam, 1990).

# Theme 2. Attitude Towards Negotiation: What is Your Attitude Towards Negotiations?

This question required a lot more trust in the relationship between interviewer and participants in order for it to be answered honestly. The interviewer believes that he was able to obtain open and honest replies. The choice of response was left to the participants own discretion.

Analysis of the data found that the attitude of most Western negotiators changed over the years working in the international business environment of Northern Iraq. The majority of participants with a Western origin preferred to achieve mutually beneficial outcomes but considered negotiation as a process in which the domestic side wins and the foreign party minimises their losses in order to maintain a working relationship. These findings are in line with the observations of Rubin and Brown (1975), who found that negotiators enter a relationship with the other party to which they bring variations of prior experience, background and expectations that may affect the manner and effectiveness of a negotiation. The findings were also in accordance with Kluckhohn and Strodtbeck's (1961) work on cross-cultural negotiations that indicated that negotiating behaviour and perceptions of individuals vary due to the underlying differences inherent in each other's culture that provide order and direction to human interaction (Clark, 1990; Kluckhohn & Strodtbeck, 1961; Limaye & Victor, 1991; McCrae, 2000). Western negotiators described their attitude towards the negotiation process in the following manner:

I must admit that my attitude changed over the years, when I arrived here my attitude was that negotiations should provide a benefit for both sides. Whereas after having been here for five years and dealing a lot with the locals I see it more as a process in which one side wins and one side loses. (Participant C, male, Western senior manager)

I would prefer to achieve a win-win situation that would be nice but you will lose something per se, it's a relationship based process. (Participant B, male, Western senior manager) In this environment you are usually not in a position of strength therefore you try to minimise your losses, as you need to maintain relationships to various degrees. (Participant A, male, Western senior manager)

By contrast, most Kurdish-Iraqi negotiators conveyed that it was important to achieve a mutually beneficial outcome, and that it is important to enter a negotiation process with a party that has substantial experience with negotiations. The data suggests that Kurdish-Iraqi participants, due to their cooperative behaviour and their sensitivity to relationships and changes, show characteristics of collectivistic cultures. This finding is in line with numerous studies that have emphasised that the relationship between negotiating parties often interacts with collectivism and individualism in activating domain-specific cultural schemas to guide the negotiator's choice (Cai et al., 2000; Wilson & Drake, 2000; Morris et al., 1998; Mintu-Wimsatt, 2002).

The following statements suggest that Kurdish-Iraqi negotiators prefer win-win outcomes that they perceive as mutually beneficial for all parties involved but they expect their counterparts to have the right mind-set to engage in negotiation with them.

You can't negotiate with uninformed people at the negotiation. They have to be informed or at least have experience with negotiations. Because a negotiation is a giving and taking, if a person doesn't have the skill or mind-set the negotiation will not be successful. (Participant E, male, Kurdish-Iraqi senior manager)

When negotiating I try always to be constructive as I don't like adversarial negotiations. (Participant G, male, Kurdish / Iraqi senior manager)

It is important to have a good relationship and that both sides get something out of it. (Participant H, male, Kurdish / Iraqi senior manager)

The general findings imply that Western and Kurdish-Iraqi negotiators are often faced with two basic strategies, problem solving or aggressive bargaining. Likewise, Kurdish-Iraqi negotiators, who aim for mutually beneficial outcomes through

interdependence with others and focus on relational harmony, show collectivist features. Whereas Western negotiators show individualistic characteristics such as a tendency to address self-serving concerns but are willing to adapt to the concessions of the other party to maintain a good relationship. In addition, the findings suggest that the attitude of Western participants towards negotiations changed during their stay in the Iraqi business environment from a win-win to a win-lose perception. Evidence suggests that the problem-solving approach is beneficial to establish and maintain long-term relationships.

# Theme 3. Methods of Communication: During Your Negotiations, How Do You Communicate With Your Opposite?

This question aimed to fathom the differences or similarities between Western and Kurdish-Iraqi negotiators with regards to their approach to communication. The interviewer was able to obtain some good replies from the respondents, reflecting the individual communication approach when negotiating.

Over the past decade numerous studies examined the meaning and context of negotiations and how culture shapes the substance of communication (Adair, 2001; Brett & Okumura, 1998; Cohen,1997; Fisher et al., 1991; Kopelman & Olekalns, 1999; Leung, 1997; Markus & Kitayama, 1991). The analysis of the findings from this study indicate that Western negotiators value directness in their communication and thereby demonstrate the characteristics of low-context communication found in individualistic cultures. This finding was consistent with several studies that suggests that low-context negotiators are likely to rely more on direct verbal strategies (Augsburger, 1992; Cohen, 1997). Similarly, communication research has supported the idea that individualistic cultures tend to stress the value of directness in their communication approach and verbalise their individual wants and needs (Trubisky et al., 1991; Kim 1994; Gudykunst et al., 1996).

Below are four extracts from the Western participants in the study.

That's another thing that changed over the years, when I arrived here I valued directness in my counterpart but this strategy has proven not to be successful. Arab culture values a more moderated approach where you talk a long time, as we say beating around the bush. You talk for an extended period about something insignificant and then maybe after half an hour of just talking you finally reach a point you would like to discuss. (Participant C, male, Western senior manager)

I like face-to-face communication but it depends on the culture, you might have to drink a little tea first, you need to warm up to the subject sometimes. However, as soon as you have warmed up to the subject you should be reasonable direct. (Participant A, male, Western senior manager)

I think in the beginning it has to be a moderated approach, when it comes to the point where both sides feel comfortable and understand that they are not there to hurt each other than you can be direct and put on the table what you want. (Participant B, male, Western senior manager)

I prefer my counterpart, I basically push it, I keep pushing it till I'm sure that they are telling me that this is going to happen. (Participant D, female, Western senior manager)

In contrast, most Kurdish-Iraqi negotiators prefer not to directly approach an issue but rather to create an environment to get to know the other person and to establish trust first. The use of explicit indirect language appears to be in line with Keegan (1989), who suggested that in high-context cultures less information is contained in verbal expression and therefore context variables such as values need to be considered to comprehend the message. This finding was also supported by negotiation literature which maintains that members of collectivist cultures tend to stress the value of contemplative talk and discretion in voicing their opinions and feelings (Gudykunst et al., 1996; Kim 1994; Trubisky et al., 1991).

The following extracts confirm that Kurdish-Iraqi negotiators demonstrate the characteristics of high-context and collectivist cultures.

When you negotiate with someone you have to try to create an environment to get through to the person. For example when you meet a French person you have to prepare a few questions over France to be polite, as you are likely to negotiate another time. You have to be polite, as you cannot be straight with that person or be in a hurry to do business. Next time you meet he will have more trust and be more faithful to you and maybe more likely to give in as you have developed a friendship or good relationship. (Participant E, male, Kurdish-Iraqi senior manager)

You ask a lot of question about unrelated topics and you are polite to your opposite; you get to know the other side and talk about other things first. (Participant E, male, Kurdish-Iraqi senior manager)

The findings imply that Western negotiators with their direct communication approach show characteristics of low-context and individualistic cultures. Therefore, Western participants rely on formal communication and that is expressed verbally with the informal context being less important in understanding the message (Simintrais & Thomas, 1998). Their low-context conflict communication aims clearly to state the speakers true intentions (Ting-Toomey & Kurogi, 1988). However, most Western negotiators, due to time spent in the region and their substantial experience in the locality, have adapted their communication approach in order to suit their local counterparts. By contrast, Kurdish-Iraqi negotiators, with their use of indirect language and contemplative talk, show characteristics of high-context and collectivistic cultures. Kurdish-Iraqis, with their indirect and high-context communication approaches, focus on concealing their true intentions (Ting-Toomey & Kurogi, 1988). Kurdish-Iraqi negotiators, as representatives of collectivist characteristics, prefer to avoid conflict and emphasise the value of maintaining relational harmony in conflict interactions (Trubisky et al., 1991).

# Theme 4. Sensitivity to Time: What Value Does Time Have For You When Negotiating?

This question caused some difficulties since, for most respondents, it was an open and indistinct question. The choice of response was left to the participant's own discretion. Many respondents talked here how they perceived time and what importance they place on time when negotiating.

A large number of studies have reported that the concept of time and value of time varies across different cultures and can have a considerable impact on cross-cultural negotiations (Bhagat & Leonard, 2002; Bluedorn et al., 1992; Foster, 1992; Hall & Hall, 1987; McDevitt, 2006; Ting-Toomey, 1999). The analysis of the interviews in this study showed that most Western participants consider time as a limited resource and prefer to use it in a linear way and thereby show traits that are associated with individualistic and monochronic societies. This finding is in accordance with Hall and Hall (1987), who suggested that monochronic cultures perceive and use time in a linear way, allowing a person to concentrate on one issue at a time. Hofstede (1993) found in his work that individualistic cultures consider time as a valuable commodity and limited resource that should be used wisely.

Some examples of replies from the Western participants follow.

You want to do it quickly but a lot of times you can't really do it. Well, it depends on the magnitude of the monetary volume. The deal I have discussed before about those \$250,000, I did it in less than 30 minutes. Unfortunately other areas might take smaller meetings over a couple of weeks; you might have to plant some seeds. I personally prefer to focus on one issue at a time. (Participant A, male, Western senior manager)

If you have to reach a critical conclusion then time is of the essence. Under normal circumstances I would connect the value of time to an expected economic outcome but that is a Western and not the local Arab approach. I would like to mention once more that time in these countries has no value. I usually like to focus on one thing but I switch to other things if I see that the process is stalled. (Participant C, male, Western senior manager)

Unfortunately in this area, time is the most important thing and you have to have a lot. You should never enter a negotiation here with the idea that you have to leave if you know that already this is never good. I prefer to focus on [one] point at a time

but if it is required I multitask and sometimes it helps you connect things you haven't thought about before. (Participant B, male, Western senior manager)

In comparison, Kurdish-Iraqi participants view time as plentiful and usually prefer to multitask. Consequently, Kurdish-Iraqi participants show traits that are associated with collectivistic and polychronic societies. This finding is in unison with the early findings of Hall and Hall (1987) who emphasise that polychronic cultures take longer to make decisions and do not discriminate among issues or prioritise them according to a Western logical flow. Polychronic cultures, through allocating a greater amount of time, put great emphasis on forming relationships rather than holding to a tight schedule (Ting-Toomey, 1999).

Below are four extracts from the Kurdish-Iraqi participants in the study to demonstrate the above findings.

I don't like time to be limited as deadlines are not good for healthy negotiations. I usually prefer to focus on several things. (Participant G, male, Kurdish-Iraqi senior manager)

It is important to take as much time as possible otherwise there is no point. I prefer to do several things, in a negotiation I want to talk about several things and if really needed we go into detail. (Participant H, male, Kurdish-Iraqi senior manager)

You have to have a lot of time for the meeting, in business you always have to plan some extra time otherwise people won't trust you. I usually prefer to multitask and to attend to many tasks. (Participant E, male, Kurdish-Iraqi senior manager)

It is important to have as much time as needed; sometimes you have several meetings over the same topic. I like to do several things at the same time. (Participant F, male, Kurdish-Iraqi senior manager)

The general findings imply that time has a direct impact on the negotiation process. Western negotiators perceive time as valuable and like to utilise and prioritise time in a linear way. Kurdish-Iraqi negotiators pay attention to all matters at the time they come up, which can cause a delay in the negotiation process. Western participants

can be associated with individualistic cultures and a monochronic perception of time, whereas Kurdish / Iraqi participants show characteristics of collectivistic cultures and embrace a polychronic perception of time (Hofstede & Usunier, 1996). As reported in numerous studies, a culture's attitude towards time determines the importance placed on the development of interpersonal relationships (Adler, 1986; Boone & Kurtz, 1996; Bovee & Thill, 1995; Foster, 1992; Hall, 1989). Kurdish / Iraqi negotiators with their boundless view of time put more emphasis on forming relationships and getting to know the other party. By contrast Western participants with *their* time consciousness place less priority on building long-term relationships.

# Theme 5. Emotionalism: When Negotiating What Is Your Opinion of Emotions at The Negotiation Table?

This question again required a high amount of trust for the respondents to open up to the interviewer. The interviewer believes that he was able to obtain open and honest replies.

Many research studies assessed the role emotions play in negotiation and reported that, due to its influence on cognition and communication, it has a practical impact on a negotiator's behaviour, the negotiation process and outcomes (Allred et al., 1997; Allred, 2002; Butt et al., 2005; Davidson & Greenhalgh, 1999; Kopelman et al., 2004; Morris & Keltner, 2000; Pillutla & Murnigham, 1996, Thompson et al., 1999). The results of this study showed that Western negotiators are less emotionally expressive. Emotional restraints are usually valued in social interactions but not often experienced in negotiations with local counterparts. These findings are in line with Mesquita and Frijida (1992) who identified, in their research on the relationship of culture and emotion, that there are large differences in the way emotion is expressed across cultures. Early research discovered that the conflict and bargaining that takes place during a negotiation has an inherent affective component and evokes emotional reactions, such as distress or anger. These are often experienced during the negotiation process (Purdy, 1967; Walton & McKersie, 1965).

Examples of responses from the Western participants on the subject include,

Once more I have to say that my approach changed over the years, when I came here first my impression [or] attitude was that emotions have no place at the negotiation table. Now after five years I have to learn that around here emotions at the negotiation table are a very common sight. People sometimes get upset, sometimes people even shout at you but these are only superficial things. (Participant C, male, Western senior manager)

There are emotions especially with people over here and you can see if they like it or not, they show that. You can also show emotions on your part but it always should be to the extent that it always leaves a backdoor. I learned here that emotions are okay, you can show strength or you could show that you are disappointed with something. Emotions are definitely there and are part of the culture here even if it is a very easy negotiation like buying something, there will be emotions. (Participant B, male, Western senior manager)

I personally don't believe that emotions are appropriate at the negotiation but some people are really good in expressing themselves. Some people go home and come back but I think it gets worse and worse as in the end something gets in there which was never meant to be in the negotiation. Emotions are definitely something you have to watch and if you realise that the other party gets too crazy you might have to postpone the meeting. (Participant D, female, Western senior manager)

Unfortunately I don't think you can disregard them but you should not put too much face in it. Like a Latino would scream and yell, typically in South America there is no win-win, they have to win. They have to feel like they kick you to death and they scream and yell, pauck and spout. It's quite interesting but you have to swallow your ego and you have to do that a lot but it's all emotions, it's all nonsense. (Participant A, male, Western senior manager)

In contrast the findings for Kurdish-Iraqi negotiators show that they are emotionally expressive and tend to value affective engagement and involvement in communicating with others. This finding is in agreement with the research of Trompenaars and Hampden-Turner (1998) who discovered that members of cultures that are emotionally expressive tend to voice and visibly demonstrate their feelings through body posture and facial expressions. Studies show that negotiators respond

to expressed anger by their counterparts in either displaying similar behaviour or demonstrating complimentary behaviour that is viewed as necessary to maintain the interaction (Butt et al., 2005). Similarly, anger may convey the impression that a negotiator will not settle for a suboptimal outcome by pressing the counterpart to be less committed to original positions but also the counterpart will lose motivation to explore creative solutions for mutual gains (Liu, 2009).

The following are examples of responses from the Kurdish-Iraqi participants.

If the other person is comfortable it has a positive effect on the negotiation but emotions do not always work. If people have negative emotions during the negotiation you should tell them that you will think about your point, tell them that they might be right and you need more time to reconsider your point and postpone the meeting to another day. Thereby you give him and yourself enough time to come up with a better solution. Emotions are an important part of the negotiation. (Participant E, male, Kurdish-Iraqi senior manager)

Emotions are important to express yourself at the negotiation table. (Participant F, male, Kurdish-Iraqi Senior Manager)

Emotions can be very helpful, especially when negotiating with emotional people. (Participant G, male, Kurdish-Iraqi senior manager)

I will try to take emotions into consideration and try to read the other party's thoughts; I will mainly focus on reading people. A negotiation is subjective and emotions are very important for it. (Participant H, male, Kurdish-Iraqi senior manager)

In general the findings for Western and Kurdish-Iraqi negotiators suggest that Western negotiators in comparison to their domestic counterparts are less emotionally expressive and value emotional restraints. Nevertheless, the majority of Western participants adapted to the domestic environment and accepted that emotions from the other party are usual when negotiating. In contrast, Kurdish-Iraqi negotiators are emotionally expressive and value emotional behaviour in the opposite party when negotiating. However, the emergence of negative emotional spirals that

cause increasingly negative emotional reactions in negotiators will hurt the negotiation process (Purdy, 1967; Walton & McKersie, 1965). There is evidence that distributive strategies including threats, positional commitments and persuasive arguments tend to yield win-lose outcomes and smaller joint profits for all parties involved (Pruit & Lewis, 1975). In comparison integrative strategies such as information exchange and relationship building tend to yield win-win outcomes, higher joint profits and greater desire to cooperate in the future (Thompson, 1990).

## Theme 6. Willingness to Take Risks: Do You Feel Comfortable in Taking Risks?

This question caused difficulties for most participants, as it is an open and rather ambiguous question.

Numerous studies reported that risk and uncertainty are generally present in the context of negotiations and differences in the perception of risk have direct implications for the exchange of negotiators (Deutsch, 1973; Markowitz, 1959; Neale & Bazerman, 1992; Ordonez et al., 2009; Warner, 1995; Weber, 1994; Weber & Hsee, 1998). The results show that Western participants are comfortable and motivated to take calculated risks if it helps to achieve their set objective. This finding is in accordance with the study of Hofstede (1993) who found that in low uncertainty-avoidance cultures individuals accept the future, are motivated by success and are more willing to take risks.

The following four extracts provide demonstrate the above finding.

I'm happy to take managed risks. Risk is fundamentally the cause of something happening, so what are the odds of it and what the consequences are. If the odds are small but the consequences are massive you need to manage it carefully, the issue is to understand the consequences. (Participant A, male, Western senior manager)

If information is not properly reported, the risk is not only not to win in a negotiation but not to establish operations in a lucrative area and you have to get

there. Either you manage or you will never get there, so yes I feel comfortable taking risks. You have to take risks, that is a part of your job and makes things more pleasant and easier but that doesn't mean that the problem is necessary solved. (Participant D, male, Western senior manager)

If you are working in countries like Iraq you always have risks and you always try to limit your risks as far as possible but this is difficult in a high risk environment like Iraq. Meaning to say that you sign a contract and after half a year later you find out that people don't feel bound to the contract anymore and all of a sudden want changes which usually involve high costs and lead to problems. You have to take a calculated risk in order to get anywhere. (Participant B, male, Western senior manager)

In contrast, the data on Kurdish-Iraqi participants showed that domestic negotiators are uncomfortable and careful to take risks and show thereby characteristics of high uncertainty-avoidance cultures. This finding is supported by the research of Hofstede (1993), who suggested that in high uncertainty-avoidance cultures individuals are motivated by a desire for security and a desire to control and predict the future. Neale and Bazerman (1992) found that an individual's cultural orientation towards risk in specific contexts tends to persist over time, forming a relatively stable pattern.

The following quotations reflect the reluctance of Kurdish-Iraqi participants to take risks.

Of course I'm not comfortable to take a risk but if it is your responsibility you might have to but I usually don't want to take a risk. (Participant H, male, Kurdish-Iraqi senior manager)

I don't feel comfortable taking risks, you always run the risk of losing when negotiating but you should try to minimise the risks. (Participant E, male, Kurdish-Iraqi senior manager)

I usually make sure to avoid taking risks. (Participant F, male, Kurdish-Iraqi senior manager)

I prefer not to take risks as you may lose business opportunities if you cannot handle the negotiation process properly. (Participant G, male, Kurdish-Iraqi senior manager)

Overall the findings imply that Western participants are willing to take calculated risks and are motivated by possible success. In comparison, their Kurdish-Iraqi counterparts prefer avoiding taking risk as they value security in their decision-making process. Douglas and Wildavsky (1983) found that each society has a general preference for certainty versus uncertainty and risk avoidance versus risk seeking. Risk-taking is a social process that is best understood through cultural analysis and interpretation. The reason for this is that each type of risk is exaggerated or minimised, feared or embraced according to the social and cultural acceptability of the underlying activities. Likewise, several studies discovered that an individual's cultural orientation towards handling risk has been found to affect negotiation processes and outcomes (Locke & Latham, 2002; Thompson, 1990)

## **Summary of Findings**

This chapter has presented the results of the analysis of the study and discusses the findings in relation to the literature review. A total of six themes and six sub-themes emerged from the analysis of the responses of the eight participants who are working in the current international business environment of Northern Iraq. The themes reflect the participants' perceptions of the cross-cultural negotiation process in the region.

The findings suggest that the negotiation goal played an important role in the negotiation process as Western and Kurdish-Iraqi negotiators both enter with motivational goals that influence how they conduct negotiations. The relationship between the two negotiating parties interacts with collectivism and individualism as a cultural schema to guide the perceived goal of a negotiation. Likewise, a negotiator's attitude tends to be an influential element in the negotiation process, as either a problem-solving or aggressive bargaining strategy is utilised. Similarly, communication emerges as a strong theme around the issue of which communication method the participants used to engage in negotiations with each other. Time is

another influential component impacting cross-cultural negotiations in the international business environment of Northern Iraq because either monochronic or polychronic perceptions of time are persistent. Finally, emotionalism is another influential factor that has a practical impact on a negotiator's behaviour, the negotiation process and outcome. The willingness of participants to take risks has a strong direct impact on the exchange between the negotiators.

The next and final chapter completes the study. It draws conclusions from the findings and summarises the findings of the research, discusses the limitations of the study, provides direction for future research and suggest implications for negotiators in the region.

## Chapter 5

### **Conclusions**

The main aim of this study was to explore the cultural differences in negotiation between Kurdish-Iraqi and Western senior managers operating in the oil industry of northern Iraqi. To achieve that purpose this study began with investigating and discussing the literature on cross-cultural negotiations with a special focus on the theoretical models of Salacuse (1991), Hall (1989) and Hofstede (2001). Following this, the researcher developed a set of questions that paid attention to the cultural dimensions that have an impact on cross-cultural negotiations. The categories were discussed in detail in the literature review.

The purpose of this study was to investigate the specific differences in negotiation between Western and Kurdish-Iraqi managers in the current international business environment of Northern Iraq. It also sought to examine the main cultural elements that influence cross-cultural negotiations. The research approach utilised a qualitative method of eight in-depth and semi-structured interviews. Grounded theory was used as a means of analysing the data obtained through the interviews. In the previous chapter the findings from the analysis were presented and discussed in comparison to the literature.

The study is based on a small sample of participants; however, it still makes two important contributions: firstly, the literature on cross-cultural negotiation in the international business environment of northern Iraq is limited and these findings add substantially to the understanding of cross-cultural negotiations in this particular environment. Secondly, it reveals that, over time, there is considerable cultural assimilation by Western participants to the local style in their negotiating behaviour. This is in terms of perception of time, the development of relationships, communication and acceptance of emotional negotiating behaviour.

One of the most obvious findings to emerge from this study was that Western and Kurdish-Iraqi negotiators enter negotiations with different motivational goals that influence their cognitive schemas (Carnevale & Probst, 1998; De Dreu & Van Kleef, 2004; De Dreu & Van Lang, 1995; Tjosvold, 1998). The relationship between the two groups can be best described as interaction of Western individualism and

Kurdish-Iraqi collectivism. This has some implications for the negotiation outcome which is somewhat expected as Kurdish-Iraqi participants are more sensitive towards establishing and maintaining a good relationship with the other party. On the other hand, Western negotiators address their negotiation needs in a self-serving manner and prefer immediate results, while the local group emphasises establishing a relationship with the other party first.

The findings also show that the negotiation attitude of the participants is influenced by variations of prior experience, background and expectations. The author believes that this may greatly affect the manner and effectiveness of a negotiation as suggested by Rubin & Brown (1975). This study shows that Kurdish-Iraqi negotiators, with their aim of mutually beneficial outcomes, show a tendency towards interdependence. They focus on relational harmony show collectivist features. Similarly, Western negotiators demonstrate individualistic features in addressing economic concerns over relational aspects, but have adapted over time to the expectations of the local negotiators in order to establish improved relationships with the aim of improving the results of their negotiations.

One of the noteworthy findings to emerge from this study is that the communication approach of Western negotiators shows characteristics of low-context and individualistic cultures. This is demonstrated by their relying on formal verbally expressed communication with the informal context being less important (Simintrais & Thomas, 1998). By contrast, Kurdish-Iraqi negotiators display characteristics of high-context communication by using indirect language and contemplative talk. The findings suggest that the low-context communication of Western participants clearly aims to state the speaker's true intentions without any disguise. This is in contrast to the high-context communication of Kurdish-Iraqi negotiators which tends to conceal the speaker's true intentions. This feature was identified by Ting-Toomey and Kurogi (1988). Western negotiators have adapted their negotiation communication approach to a mix of contemplative talk to suit the local party and direct communication to address their needs.

Another major finding was that Kurdish-Iraqi and Western negotiators differ in the degree of emotional expressiveness and emotional restraints. Mesquita and Frijida (1992), observed that this behaviour is common, and even valued and practiced in

the Kurdish environment. Western negotiators, in comparison to their domestic counterparts, are less emotionally expressive and value emotional restraints. However, the majority of Western participants has adapted to the local environment and accepts that emotions from the other party are part of the domestic negotiation practices.

This study found that the concept of time varies considerably in different cultures. For both the Kurdish-Iraqi and the Western participants this has an effect on how they conduct their cross-cultural negotiations. This is in line with the findings of Bhagat & Leonard (2002), Bluedorn et al. (1992), Foster (1992), Hall and Hall (1987), McDevitt (2006 and Ting-Toomey (1999). Western participants perceive time and prefer to prioritise time in a linear way and thereby show individualistic features and a monochronic perception of time during negotiations. Domestic negotiators embrace a polychronic attitude towards time and pay attention to all matters at the time they come up and delay the negotiation process. This is in line with Hofstede and Usunier (1996). The Kurdish-Iraqi negotiators acted as if they had boundless amount of time and put more emphasis on forming relationships and getting to know the other party.

The findings identified that risk and uncertainty in the context of negotiations have direct implications for the exchanges between negotiators and the negotiation outcome. This is in line with the findings of Deutsch (1973), Markowitz (1959), Neale and Bazerman (1992), Ordonez et al. (2009), Warner (1995), Weber (1994) and Weber and Hsee (1998). Western participants are willing to take calculated risks to advance their negotiation outcome and show increased contentious behaviour. Kurdish-Iraqi negotiators avoid risks and value security in their decision-making process and display increased concessionary behaviour in their negotiation approach.

## **Limitations of the Study**

Like every study, this study has inherent limitations. This study involved only eight participants from a Western and Kurdish-Iraqi background. Although these eight freely shared their experiences with regards to cross-cultural negotiations in the current international business environment of northern Iraq, the small number of

participants and the focus on the northern region of Iraq may limit the extent to which the findings can be generalised.

English was used as a means of communication. English is a second language for the majority of the participants who all have different mother languages. However, all participants use English as the *lingua franca* in their profession and are very capable of expressing and sharing their experiences. It has to be acknowledged that there is a possibility that the participants may have expressed their views differently in their own mother tongue.

## **Implications and Recommendations for Future Research**

Despite these limitations, this research provides a contribution to the investigation of cross-cultural negotiations. Thereby, this research will serve as a good starting point to further investigate cross-cultural negotiations in this region. Given the limited empirical data in the field of intercultural negotiation in Iraq, this study contributes to the pool of knowledge and provides valuable insights into the local intercultural negotiation practices. The findings of this study have implications for local negotiation practices. These implications centre on shaping expectations about the differences in negotiation approaches of Western and local negotiators when negotiating in the current Kurdish-Iraqi international business environment. Negotiators need to be aware of the relational aspects as well as the likely approach to managing conflict. Assimilating to the local negotiation style can build common bonds and provide indirect ways of signalling interest and goodwill as a means of facilitating beneficial outcomes for all parties involved. It is important to investigate further how negotiations in this international business environment are conducted. However, one cannot generalise or presume that the results of this study apply to other industries or non-Kurdish regions in Iraq.

Future research should focus on comparing the differences between Kurdish-Iraqi and Western cross-cultural negotiators with participants from other industries or Arab regions. This will help to add more significance to the study and expand the knowledge on the cultural factors that influence cross-cultural negotiation in the region. The scope for future research could be extended. For instance, research

could be conducted as longitudinal studies with selected senior managers to observe how their negotiation strategies adapt and change over time. It is recommended that the researcher should use control groups with local negotiators negotiating with participants from an Arab Sunni or Shi'a background in order to make comparisons with cross-cultural negotiations to identify different negotiation processes and strategies. It is hoped that the research will improve the negotiation practice and lead to a better understanding across the cultures in this region.

### **Reference List**

Adair, W. L., Brett, J., Lempereur, A., Okumura, T., Shikhirev, P., Tinsley, C., & Lytle, A. (2003). Culture and negotiation strategy. *Negotiation Journal*, 20(1), 87–111. doi: 10.1111/j.1571-9979.2004.00008.x

Adair, W. L. (2000). *Reciprocity in the global market: Cross-cultural negotiations* (Doctoralxdissertation).xRetrieveddfrom:

http://ezproxy.aut.ac.nz/login?url=http://search.proquest.com/docview/304612217?accountid=8440

Adler, N. J. (1986). International Dimensions of Organizational Behavior. *The International Executive*, 28(1), 31.

Adler, N. J. (2002). Competitive frontiers: Cross-cultural management and the 21st century. *International Journal of Intercultural Relations*, 19(4), 523–537. doi:/10.1016/0147-1767(95)00032-1

Adler, N. J. (2002). Global Companies, Global Society. *Journal of Management Inquiry*, 11(3), 255–260. doi: 10.1177/1056492602113008

Adler, N. J., Graham, J. L. (1998). Business negotiations in Canada, Mexico, and the United States. *The International Executive*, *30*(1), 13. doi:10.1002/tie.5060300104

Allred, K. (2000). Distinguishing best and strategic practices: A framework for managing the dilemma between claiming and creating value. *Negotiation Journal*, *16*(4), 387–397. doi: 10.1023/A:1026640906076

Allred, K. G., Mallozzi, J. S., Matsui, F., & Raia, C. P. (1997). The influence of anger and compassion on negotiation performance. *Organizational Behavior and Human Decision Processes*, 70(3), 175–187. doi:10.1006/obhd.1997.2705

Ashforth, B. E., & Mael, F. (1989). Social identity theory and the organisation. *Academy of Management Review*, 14, 20-39.

Augsburger, D. W. (1992). *Conflict mediation across cultures: pathways and patterns*. Louisville, KY: Westminster/ John Knox Press.

Barry, B. F., Friedman, R. (1998). Bargainer characteristics in distributive and integrative negotiation. *Journal of Personality and Social Psychology*, 74(2), 345–359. Retrieved from:

http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s.q=Bargainer+characteristics+in+distributive+and+integrative+negotiation

Bhagat, R. S., & Leonard, K. M. (2002). How non-Americans view American use of time: A cross-cultural perspective. In P. Boski (Ed.). *New directions in cross-cutural psychology* (pp.183–191). Warsaw, Poland: Polish Psychological Association.

Bluedorn, A. C., Kaufman, C. F., & Lane, P. M. (1992). How many things do you like to do at once? An introduction to monochronic and polychronic time. *Academy of Management Executive*, *6*(4), 17–26. Retrieved from: http://search.proquest.com.ezproxy.aut.ac.nz/docview/210524065

Bond, M. H., & Luk, C. L. (1993). Personality variation and values endorsement in Chinese university students. *Personality and Individual Differences*, *14*(3), 429-437. doi:10.1016/0191-8869(93)90312-Q

Boone, L. E., & Kurtz, D. L. (1996). *Contemporary business*. Fort Worth, TX: Dryden Press.

Bovée, C. L., & Thill, J. V. (1995). *Marketing (2nd ed.)*. New York, NY: McGraw-Hill.

Brett, J. M. (2000). Culture and Negotiation. *International Journal of Psychology*, 35(2), 97–104. Retrieved from:

http://isites.harvard.edu/fs/docs/icb.topic551848.files/Brett.pdf

Brett, J. M., & Okumura, T. (1998). Inter- and intracultural negotiation: U.S. and Japanese negotiators. *The Academy of Management Journal*, 41(5), 495–510. Retrieved from: http://www.jstor.org.ezproxy.aut.ac.nz/stable/256938

Brett, J. M., & Okura, I. (2000). Culture and Negotiation. *International Journal of Psychology*, 35(2), 97–104. Retrieved from:

http://isites.harvard.edu/fs/docs/icb.topic551848.files/Brett.pdf

Brockhaus, R. H. (1980). *The psychology of the entrepreneur*. Englewood Cliffs, NJ: Prentice Hall.

Bryman, A., & Bell, E. (2007). *Business Research Methods* (3rd ed.). Oxford, UK: Oxford University Press.

Butt, A. N., Choi, J. N., & Jaeger, A. M. (2005). The effects of self-emotion, counterpart emotion, and counterpart behavior on negotiator behavior: A comparison of individual-level and dyad-level dynamics. *Journal of Organizational Behaviour*, 26,x681–704. Retrieved from:

http://onlinelibrary.wiley.com.ezproxy.aut.ac.nz/doi/10.1002/job.328/abstract

Buttery, E. A., & Leung, T. K. P. (1998). The difference between Chinese and Western negotiations. *European Journaldof Marketing*, 32(4), 374–389 doi: 10.1108/03090569810204652

Cai, A., Wilson, R., & Drake, E. (2000). Culture in the context of intercultural negotiation. *Human Communication Research*, 26(4), 591–617. doi: 10.1111/j.1468-2958.2000.tb00770.x

Cai, D. A., & Drake, L. E. (1998). The business of business negotiation. In M.E. Roloff (Ed.), *Communication Yearbook* (Vol. 21, pp. 153–189). Thousand Oaks, CA: SAGE Publications.

Carlson, J. A., & Davis, C. M. (1971). Cultural values and the risky shift: A cross-cultural test in Uganda and the United States. *Journal of Personality and Social Psychology*, 20(3), 392–399. doi:10.1037/h0031906

Carnevale, P. J., & Probst, T. M. (1998). Social values and social conflict in creative problem solving and categorization. *Journal of Personality and Social Psychology*, 74(5), 1300–1309.

Retrieved from: http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s. q=Social+values+and+social+conflict+in+creative+problem+solving+and+categoriz ation

Carnevale, P. J. D., & Isen, A. M. (1986). The influence of positive affect and visual access on the discovery of integrative solutions in bilateral negotiation.

Organizational Behavior and Human Decision Processes, 37(1), 1–13. doi:10.1016/0749-5978(86)90041-5

Charmaz, K. (2006). Constructing grounded theory: A practical guide through qualitative analysis (1st ed.). Thousand Oaks, CA: SAGE Publications.

Clark, T. (1990). International marketing and national character: A review and proposal for an integrative theory. *Journal of Marketing*, *54*(4), 66–79. doi: 10.2307/1251760

Cohen, A. (1997). Personal and organizational responses to work-nonwork interface as related to organizational commitment. *Journal of Applied Social Psychology*, 27(12), 1085–1114. doi: 10.1111/j.1559-1816.1997.tb00288.x

Creswell, J. W. (2003). Research design: Qualitative, quantitative, and mixed method approaches (2nd ed.). Thousand Oaks, CA: SAGE Publications.

Creswell, J. W. (2013). Qualitative inquiry and research design: Choosing among five approaches (3rd ed.). Thousand Oaks, CA: SAGE Publications.

Cross, J. G. (1977). Negotiation as a learning process. *The Journal of Conflict Resolution*, 21(4), 581–606. doi: 10.2307/173613

Davidson, M. N., & Greenhalgh, L.(1999). The role of emotion in negotiation: The impact of anger and race. In R. J. Bies, R. J. Lewicki, & B. H. Sheppard (Eds.), *Research on negotiation in organizations* (pp. 3–26) Greenwich, CT: JAI Press.

De Dreu, C. K., Van Kleef, G. A., & Manstead, K. W. (2004). The interpersonal effects of emotions in negotiations: A motivated information processing approach. *Journal of Personality and Social Psychology*, 87(4), 510–528. Retrieved from: http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s.q=The+interpersonal+effects+of+emotions+in+negotiations%3A+A+motivated+information+processing+approach

De Dreu, C. K. W., & van Lange, P. A. M. (1995). The impact of social value orientations on negotiator cognition and behavior. *Personality and Social Psychology Bulletin*, 21(11), 1178–1188. doi: 10.1177/01461672952111006

Deal, T. E., & Kennedy, A. A. (1982). *Corporate cultures: The rites and rituals of corporate life*. New York, NY: Perseus Books Publishing.

Deutsch, M. (1973). The resolution of conflict: Constructive and destructive processes. New Haven, CT: Yale University Press.

Dick, B. (2002). Postgraduate programs using action research. *The Learning Organization*, 9(4), 159–170. doi:10.1108/09696470210428886

Douglas, M., & Wildavsky, A. (1982). Risk and culture: An essay on selection of technological and environmental dangers. Berkley, CA: University of California Press.

Drake, L. E. (2001). The culture-negotiation link: Integrative and distributive bargaining through an intercultural communication lens. *Human Communication Research*, 27(3), 317–349. doi: 10.1111/j.1468-2958.2001.tb00784.x

Farber, H. S., & Katz, H. C. (1979). Interest arbitration, outcomes, and the incentive to bargain. *Industrial and Labor Relations Review*, 33(1), 55–63. doi: 10.2307/2523045

Fisher, R. Ury, W. L., & Patton, B.M. (1991). *Getting to yes: Negotiating agreement without giving in*. New York, NY: Penguin.

Foster, P. (1992). Equal treatment and cultural difference in multi-ethnic schools: A critique of the teacher ethnocentrism theory. *International Studies in Sociology of Education*, 2(1), 89–103. doi: 10.1080/0962021920020106

Gelfand, M. J., & Christakopoulou, S. (1999). Culture and negotiator cognition: Judgment accuracy and negotiation processes in individualistic and collectivistic cultures. *Organizational Behavior and Human Decision Processes*, 79(3), 248–269. doi:10.1006/obhd.1999.2845

Gelfand, M. J., & Realo, A. (1999). Individualism-collectivism and accountability in intergroup negotiations. *Journal of Applied Psychology*, 84(5), 721–736. doi: 10.1037/0021-9010.84.5.721

Ghauri, P. N., & Usunier, J.-C. (Eds.). (1996). *International business negotiations*. Oxford, UK: Pergamon.

Glaser, B. G. (1978). Theoretical sensitivity: Advances in the methodology of grounded theory. Mill Valley, CA: Sociology Press.

Glaser, B. G. (2005). *The grounded theory perspective III: Theoretical coding*. Mill Valley, CA: The Sociology Press.

Goering, E. M. (1997). Integration versus distribution in contract negotiations: An interaction analysis of strategy use. *Journal of Business Communication*, *34*(4), 383–400. doi: 10.1177/002194369703400404

Graham, J. L. (1985). The influence of culture on the process of business negotiations: An exploratory study. *Journal of International Business Studies*, 16(1), 81–96. Retrieved from:

http://www.jstor.org.ezproxy.aut.ac.nz/stable/154485Graham, J. L., Mintu, A. T., & Rodgers,

Greenhalgh, L. N., S. (1985). *Determining outcomes of negotiation: An empirical assessment*. Beverly Hills, CA: SAGE Publications.

Gudykunst, W. B., Gao, G., Schmidt, K. L., Nishida, T., Bond, M. H., Leung, K., Barraclough, R. A. (1992). The influence of individualism collectivism, self-monitoring, and predicted-outcome value on communication in ingroup and outgroup relationships. *Journal of Cross-Cultural Psychology*, 23(2), 196–213. doi: 10.1177/0022022192232005

Gudykunst, W. B., & Kim, Y. Y. (1992). Communicating with strangers: An approach to intercultural communication (3rd edition). Boston, MA: McGraw-Hill.

Gudykunst, W. B., Matsumoto, Y., Ting-Toomey, S., Nishida, T., Kim, K., & Heyman, S. (1996). The influence of cultural individualism-collectivism, self-construas, and individual values on communication styles across cultures. *Human Communication Research*, 22(4), 510–543. doi: 10.1111/j.1468-2958.1996.tb00377.x

Hall, E. T. (1989). Learning the Arabs silent language. *Psychology Today*, 45–52.

Hall, E. T., & Hall, M. D. (1987). *Hidden differences: Doing business with the Japanese*. New York, NY: Anchor.

Heath, C., Larrick, R. P., & Wu, G. (1999). Goals as reference points. *Cognitive Psychology*, *38*(1), 79–109. doi:10.1006/cogp.1998.0708

Herman, M. G., & Kogan, N. (1977). Effects of negotiators personalities on negotiating behaviour. In D. Druckman (Ed.), *Negotiations: Social-psychological perspectives*. Beverly Hills, CA: SAGE Publications.

Hofstede, G. (1980). Cultures consequences: International differences in work-related values. Newbury Park, CA: SAGE Publications.

Hofstede, G. (1993). Cultural constraints in management theories. *The Executive*, 7(1), 81–94.

Hofstede, G., & Bond, M. H. (1984). Hofstede's culture dimensions: An independent validation using Rokeach's Value Survey. *Journal of Cross-Cultural Psychology*, 15(4), 417–433. doi:10.1177/0022002184015004003

Hofstede, G., & Usunier, J.-C. (1996). Hofstede's dimensions of culture and their influence on international business negotiations. In P. N. Ghauri & J.-C. Usunier, *International business negotiations* (pp. 119–129). Oxford, UK: Pergamon.

Hofstede, G. H. (1984). Culture's consequences: international differences in work-related values. Newbury Park, CA: SAGE Publications.

Hofstede, G. H. (2001). Culture's consequences: Comparing values, behaviors, institutions, and organizations across nations. Thousand Oaks, CA: SAGE Publications.

Hunter, J. E. (1996). Cognitive ability, cognitive aptitudes, job knowledge, and job performance. *Journal of Vocational Behavior*, 29(3), 340–362. doi: 10.1016/0001-8791(86)90013-8

International Energy Agency (IEA). (2012). *Iraq energy outlook*. Retrieved from: http://www.worldenergyoutlook.org/iraq/

Kahn, R. L., & Cannell, C. F. (1957). *The dynamics of interviewing; theory, technique, and cases.* Oxford, UK: John Wiley & Sons.

Keck, K. L., & Samp, J. A. (2007). The dynamic nature of goals and message production as revealed in a sequential analysis of conflict interactions. *Human Communication Research*, *33*(1), 27–47. doi: 10.1111/j.1468-2958.2007.00287.x

Keegan, W.J. (1989). *Global marketing management*. Englewood Cliffs, NJ: Prentice-Hall.

Keltner, D., & Haidt, J. (1999). Social functions of emotions at four levels of analysis. *Cognition & Emotion*, 13(5), 505–521. doi:10.1080/026999399379168

Kim, U. (1994). Individualism and collectivism: Conceptual clarification and elaboration. In U. Kim et al. (Eds.), *Individualism and collectivism: Theory, method, and applications* (pp. 19–40). Newbury Park, CA: SAGE Publications.

Kim, M.-S., Shin, H.-C., & Cai, D. (1998). Cultural influences on the preferred forms of requesting and re-requesting. *Communication Monographs*, 65(1), 47–66. doi: 10.1080/03637759809376434

Kimmel, P. R. (1994). Cultural perspectives on international negotiations. *Journal of Social Issues*, *50*(1), 179–196.

Kluckhohn, F. R., & Strodtbeck, F. L. (1961). *Variations in value orientations*. Evanston, IL: Row, Peterson.

Kopelman, S., & Olekalns, M. (1999). Process in cross-cultural negotiations. *Negotiation Journal*, *15*(4), 373–380. doi: 10.1023/A:1007557922494

Kopelman, S., Rosette, A. S., & Thompson, L. (2004). The three faces of Eve: Strategic displays of positive, negative, and neutral emotions in negotiations. *Organizational Behavior and Human Decision Processes*, 99(1), 81–101. doi:10.1016/j.obhdp.2005.08.003

Leung, K. (1997). Negotiation and reward allocations across cultures. In P. C. Early & M. Erez (Eds.), *New perspectives on international/organizational psychology* (pp.640–675). San Francisco, CA Jossey-Bass.

Lewicki, R. J., Saunders, D. M., Barry, B., & Friedman, R. A (2006). *Negotiations* (5th ed.). Boston, MA: McGraw-Hill/Irwin.

Lewicki, R. J., Saunders, D. M., & Minton, J. W. (1994). *Essentials of negotiation* (2nd ed.). Boston, MA: McGraw-Hill/Irwin.

Limaye, M. R., & Victor, D. A. (1991). Cross-cultural business communication research: State of the art and hypotheses for the 1990s. *Journal of Business Communication*, 28(3), 277–299. doi: 10.1177/002194369102800306

Liu, L. A., Friedman, R. A., & Chi, S.-C. (2005). 'Ren Qing' versus the 'Big Five': The role of culturally sensitive measures of individual difference in distributive negotiations. *Management and Organization Review*, 1(2), 225-

247. Retrieved from:

http://onlinelibrary.wiley.com.ezproxy.aut.ac.nz/doi/10.1111/j.1740-8784.2005.00010.x/abstract

Liu, M. (2008). The intrapersonal and interpersonal effects of anger on negotiation strategies: A cross-cultural investigation. *Human Communication Research*, *35*(1), 148–169. doi: 10.1111/j.1468-2958.2008.01342.x

Locke, E. A., & Latham, G. P. (2002). Building a practically useful theory of goal setting and task motivation: A 35 year odyssey. *American Psychologist*, *57*(9), 705–717. Retrieved from:

 $http://aut.summon.serials solutions.com.ezproxy.aut.ac.nz/search?s.q=Building+a+pr\\ actically+useful+theory+of+goal+setting+and+task+motivation\%3A+A+35+year+o\\ dyssey$ 

Manrai, L. A., & Manrai, A. K. (2012). Current issues in the cross-cultural and cross-national consumer research in the twenty-first century. *Journal of International Consumer Marketing*, 23(3–4), 167–180. doi: 10.1080/08961530.2011.578056

March, J. G., & Shapira, Z. (1992). Variable risk preferences and the focus of attention. *Psychological Review*, 99(1), 172–183.

Markowitz, H. M. (1959). *Portfolio selection: Efficient diversification of investments*. New York, NY: John Wiley \$ Sons.

Markus, H. R., & Kitayama, S. (1991). Culture and the self: Implications for cognition, emotion, and motivation. *Psychological Review*, 98(2), 224–253. Retrieved from:

http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s.q=Culture+and+t he+self%3A+Implications+for+cognition%2C+emotion%2C+and+motivation

Marr, P. (2012). The modern history of Iraq (3rd ed.). Boulder, CO: Westview Press.

Masuda, T., & Nisbett, R. E. (2001). Attending holistically versus analytically: Comparing the context sensitivity of Japanese and Americans. *Journal of Personality and Social Psychology*, 81(5), 922–934.

McCrae, R. R. (2000). Trait psychology and the revival of personality and culture studies. *American Behavioral Scientist*, 44(1), 10–31. doi: 10.1177/00027640021956062

McDevitt, A. S. (2006). Cultural variations as predictors of negotiation outcomes: Implications for international business communication. (Doctoral thesis). Memphis, TN: University of Memphis, Retrieved from: http://ezproxy.aut.ac.nz/login?url=http://search.proquest.com/docview/304907565?a ccountid=8440

Meina, L. (2011). Cultural differences in goal-directed interaction patterns in negotiation. *Negotiation and Conflict Management Research*, *4*(3), 178–199. doi: 10.1111/j.1750-4716.2011.00079.x

Meina, L., & Wilson, S. R. (2011). The effects of interaction goals on negotiation tactics and outcomes: A dyad-level analysis across two cultures. *Communication Research*, 38(2), 248–277. doi: 10.1177/0093650210362680

Mesquita, B., & Frijda, N. H. (1992). Cultural variations in emotion: A review. *Psychological Bulletin*, 112(2), 179–204. Retrieved from:

http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s.q=Cultural+variations+in+emotion%3A+A+review

Miles, M. B., & Huberman, A. M. (1994). *Qualitative data analysis* (2nd ed.). London, UK: SAGE Publications.

Miller, G., Boster, F., Roloff, M., & Seibold, D. (1977). Compliance-gaining message strategies: A typology and some findings concerning effects of situational

differences. *Communications Monographs*, 44(1), 37–51. doi: 10.1080/03637757709390113

Mintu-Wimsatt, A. (2002). Personality and negotiation style: The moderating effects of cultural context. *Thunderbird International Business Review*, 44(6), 729–748. doi:10.1002/tie.10047

Morris, M. W., & Fu, H.-Y. (2001). How does culture influence conflict resolution? A dynamic constructivist analysis. *Social Cognition*, *19*, 324–349. doi: 10.1521/soco.19.3.324.21475

Morris, M. W., & Keltner, D. (2000). How emotions work: The social functions of emotional expression in negotiations. *Research in Organizational Behavior*, 22, 1–50. doi: 10.1016/S0191-3085(00)22002-9

Morris, M. W., Williams, K. Y., Leung, K., Larrick, R., Mendoza, M. T., Bhatnagar, D., Hu, J.-C. (1998). Conflict management style: Accounting for cross-national differences. *Journal of International Business Studies*, 29(4), 729–747.doi:10.1057/palgrave.jibs.8490050

Morse, J. M. (1999). The case of the missing data. *Qualitative Health Research*, 21(9), 1163–1164. doi: 10.1177/1049732311410359

Myers, M. D. (1997). Qualitative research in information systems. *MIS Quarterly*, 21(2), 241–242. doi: 10.2307/249422

Myers, M. D. (2009). Qualitative research in business and management. London, UK: SAGE Publications.

Neale, M. A., & Bazerman, M. H. (1985). The effects of framing and negotiator overconfidence on bargaining behaviors and outcomes. *The Academy of Management Journal*, 28(1), 34–49. doi: 10.2307/256060

Neale, M. A., & Bazerman, M. H. (1992). Negotiating rationally: The power and impact of the negotiator's frame. *The Executive*, *6*(3), 42–51. Retrieved from: http://search.proquest.com.ezproxy.aut.ac.nz/docview/210534217

Neale, M. A., & Bazerman, M. H. (1992). Negotiator cognition and rationality: A behavioral decision theory perspective. *Organizational Behavior and Human Decision Processes*, 51(2), 157–175. doi: 10.1016/0749-5978(92)90009-v

Neale, M. A., & Northcraft, G. B. (1989). Behavioral negotiation theory: A framework for conceptualizing dyadic bargaining. Evanston, IL: Northwestern University.

Ober, S. (1995). The bulletin is dead. Long live the quarterly! *Business Communication Quarterly*, 58(1), 3–4. doi: 10.1177/108056999505800102

Ordóñez, L. D., Schweitzer, M. E., Galinsky, A. D., & Bazerman, M. H. (2009). Goals gone wild: The systematic side effects of overprescribing goal setting. *Academy of Management Perspectives*, 23(1), 6–16. doi: 10.5465/AMP.2009.37007999

Parnell, J. A., & Kedia, B. L.(1996). The impact of national culture on negotiating behaviors across borders. *International Journal of Value-Based Management*, *9*(1), 45–61.

Perdue, B. C., & Summers, J. O. (1991). Purchasing agents' use of negotiation strategies. *Journal of MarketingxResearch*, 28(2),x175–189. http://www.jstor.org/stable/3172806

Purdy, L. R. (1967) Organizational conflict: Concepts and models. *Administrative Science Quarterly*, 12(2), 296–320. doi: 10.2307/2391553

Pruitt, D. G., & Carnevale, P.J. (1993). *Negotiation in social conflict*. Melbourne, Australia: Brooks/Cole.

Pruitt, D. G., & Lewis, S. A. (1975). Development of integrative solutions in bilateral negotiation. *Journal of Personality and Social Psychology*, *31*(4), 621–633. doi: 10.1037/0022-3514.31.4.621

Pruitt, D. G., & Teger, A. I. (1969). The risky shift in group betting. *Journal of Experimental Social Psychology*, 5(2), 115–126. doi: 10.1016/0022-1031(69)90041-9

Putnam, L. L., & Jones, T. S. (1982). Reciprocity in negotiations: An analysis of bargaining interaction. *Communication Monographs*, 49(3), 171–191. doi: 10.1080/03637758209376080

Putnam, L. L., & Roloff, M. E. (1992). *Communication and negotiation*. Newbury Park, CA: SAGE Publications.

Rahim, M. A., & Magner, N. R. (1995). Confirmatory factor analysis of the styles of handling interpersonal conflict: First-order factor model and its invariance across groups. *Journal of Applied Psychology*, 80(1), 122–132. Retrieved from: http://web.ebscohost.com.ezproxy.aut.ac.nz/ehost/pdfviewer/pdfviewer?sid=81f5a45 2-af0a-4718-bdde-bd8b0e106ac6%40sessionmgr111&vid=2&hid=114

Remenyi, D., Williams, B., Money, A., & Swartz, E. (1998). *Doing research in business and management: An introduction to process and method. London, UK:* SAGE Publications.

Remmen, D. (2003). Performance pays off. *Strategic Finance*. Retrieved from http://www.imanet.org/PDFs/Public/SF/2003\_03/0303remmen.pdf

Robson, C. (2002). Real world research: A resource for social scientists and practitioner-researchers (2nd ed.): Chichester, UK: John Wiley & Sons.

Rubin, J. Z., & Brown, B. R. (1975). The social psychology of bargaining and negotiation.

Salacuse, J. W. (1991). *Making global deals: Negotiating in the international marketplace*. Boston, MA: Houghton Mifflin.

Salacuse, J. W. (1999). Intercultural negotiation in international business. *Group Decision and Negotiation*, 8(3), 217–236. doi: 10.1023/A:1008660330550

Saunders, M., Lewis, P., & Thornhill, A. (2000). *Research methods for business students* (2nd ed.). Essex, UK: Pearson Education.

Saunders, M., Lewis, P., & Thornhill, A. (2009). *Research methods for the business student* (5th ed.). Essex, UK: Pearson Education.

Sawyer, J., & Guetzkow, H. (1965). *Bargaining and negotiation in international relations*. Orlando, FL: Holt, Rinehart and Winston.

Sherman, R. & Kiresuk, T. J. (1968). Goal attainment scaling: A general method for evaluating comprehensive community health programs. *Community Mental Health Journal*, *4*(6), 443–453. doi: 10.1007/BF01530764

Shure, G. H., & Meeker, R. J. (1967). A personality/attitude schedule for use in experimental bargaining studies. *The Journal of Psychology*, 65(2), 233–252. doi: 10.1080/00223980.1967.10544868

Sinaceur, M., & Tiedens, L. Z. (2006). Get mad and get more than even: When and why anger expression is effective in negotiations. *Journal of Experimental Social Psychology*, 42(3), 314–322. doi: 10.1016/j.jesp.2005.05.002

Sitkin, S. B., & Pablo, A. L. (1992). Reconceptualizing the determinants of risk behavior. *Academy of Management Review*, 17(1), 9–38. doi: 10.5465/amr.1992.4279564

Slovic, P. (1964). Assessment of risk taking behavior. *Psychological Bulletin*, 61(3), 220–233. doi: 10.1037/h0043608

Stoner, J. A. F. (1968). Risky and cautious shifts in group decisions: The influence of widely held values. *Journal of Experimental Social Psychology*, 4(4), 442–459. doi: 10.1016/0022-1031(68)90069-3

Terhune, K. W. (1970). The effects of personality in cooperation and conflict. In P. Swingle (Ed.), *The structure of conflict* (pp. 193–234). New York, NY: Academic Press.

The Economist. (2013, 2 March). Iraq ten years on: The slow road back. Retrieved from: http://www.economist.com/news/briefing/21572777-decade-after-american-led-invasion-iraq-place-patchy-progress-and-dysfunctional

Thompson, L. (1990). Negotiation behavior and outcomes: Empirical evidence and theoretical issues. *Psychological Bulletin*, *108*(3), 515–532. Retrieved from: http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s.q=Negotiation+b ehavior+and+outcomes%3A+Empirical+evidence+and+theoretical+issues

Thompson, L. (1998). A new look at social cognition in groups. *Basic and Applied Social Psychology*, 20(1), 3–5. doi: 10.1207/s15324834basp2001\_1

Thompson, L. N., Nadler, J., & Kim, P. H. (1999). Some like it hot: The case for the emotional negotiator. In L. L. Thompson, J. M. Levine & D. M. Messick (Eds.), *Shared cognition in organisations: The management of knowledge* (pp. 139–162). Mahwah, NJ: Lawrence Erlbaum Assocs.

Ting-Toomey, S. (1999). *Communicating across cultures*. New York, NY: Guilford Press.

Ting-Toomey, S., & Kurogi, A. (1988). Facework competence in intercultural conflict: An updated face-negotiation theory. *International Journal of Intercultural Relations*, 22(2), 187–225. doi: 10.1016/S0147-1767(98)00004-2

Tjosvold, D. (1998). Cooperative and competitive goal approach to conflict: Accomplishments and challenges. *Applied Psychology*, 47(3), 285-313. doi: 10.1111/j.1464-0597.1998.tb00025.x

Trompenaars, F., & Hampden-Turner, C. (1998). *Riding the waves of culture: Understanding diversity in global business*. New York, NY: McGraw-Hill.

Trubisky, P., Ting-Toomey, S., & Lin, S.-L. (1991). The influence of individualism-collectivism and self-monitoring on conflict styles. *International Journal of Intercultural Relations*, *15*(1), 65–84. doi: 10.1016/0147-1767(91)90074-Q

Tutzauer, F., & Roloff, M. E. (1988). Communication processes leading to integrative agreements: Three paths to joint benefits. *Communication Research*, 15(4), 360–380. doi: 10.1177/009365088015004002

Urquhart, C. (2001). An encounter with grounded theory: Tackling the practical and philosophical issues. In E. M. Trauth (Ed.), *Qualitative research in IS: Issues and trends* (pp. 104–140). Hershey, PA: Idea Group Publishing.

Usunier, J.-C. (1996). Cultural aspects of international business negotiations. In P.N. Ghauri & J.-C. Usunier (Eds.) *International business negotiations* (pp. 91–118). Oxford, UK: Pergamon.

Van Kleef, G. A., & De Dreu, C. K. W. (2004). The influence of power on the information search, impression formation, and demands in negotiation. *Journal of Experimental Social Psychology*, 40(3), 303–319. doi: 10.1016/j.jesp.2003.07.004

Wall, J. A., & Blum, M. W. (1991). Negotiations. *Journal of Management Inquiry*, 17(2), 273–303. doi: 10.1177/014920639101700203

Wallach, M. A., & Kogan, N. (1965). The roles of information, discussion, and consensus in group risk taking. *Journal of Experimental Social Psychology, 1*(1), 1–19. Doi:10.1016/0022-1031(65)90034-X

Walt, V.. (2013, May, 20th). The next oil boom. Fortune, pp. 82–88.

Walton, R. E. & McKersie, R. B. (1965). A behavioral theory of labor negotiations. New York, NY: McGraw-Hill.

Warner, J. (1995, 30 October). The world is not always your oyster: Why cross-border mergers so often come to grief. *Business Week*. Retrieved from: http://www.businessweek.com/stories/1995-10-29/the-world-is-not-always-your-oyster

Watson, R., & Myers, M. D. (2001). IT industry success in small countries: The cases of Finland and New Zealand. *Journal of Global Information Management* (*JGIM*), 9(2), 4–14. doi: 10.4018/jgim.2001040101

Weber, E. U. (1993). From subjective probabilities to decision weights: The effect of asymmetric loss functions on the evaluation of uncertain outcomes and events. *Psychological Bulletin*, 115(2), 228–242. Retrieved from:

http://aut.summon.serialssolutions.com.ezproxy.aut.ac.nz/search?s.q=From+subjective+probabilities+to+decision+weights%3A+The+effect+of+asymmetric+loss+functions+on+the+evaluation+of+uncertain+outcomes+and+events

Weber, E. U., & Hsee, C. K. (1998). Cross-cultural differences in risk perception, but cross-cultural similarities in attitudes towards perceived risk. *Management Science*, 44(9), 1205–1217. doi: 10.2307/2634710

Weiss, S. E. (1994). Negotiating with "Romans" – Part 1. *MIT Sloan Management Review*. Retrieved from:

http://search.proquest.com.ezproxy.aut.ac.nz/docview/224959314

Westbrook, K. W., Ghosh, P., & Dev, D. (1996). Risk coordinative maneuvres during buyer-seller negotiations. *Industrial Marketing Management*, 25(4), 283–292. doi: 10.1016/0019-8501(95)00130-1

Wilson, S. R., & Putnam, L. L. (1990). Interaction goals in negotiation. In J. A. Anderson (Ed.), *Communication Yearbook* (pp. 374–406).Oxford, UK: Routtledge.

Wuest, J. (2007). Grounded theory: The method. In P. L. Munhall (Ed.), *Nursing research: A qualitative perspective* (pp. 225–256). Sudbury, MA: Jones & Bartlett.

# **Appendix A: Participant Information Sheet**

# Participant Information Sheet



**Date Information Sheet Produced: 22/05/2013** 

#### **Project Title**

Managing cross-cultural negotiations in the current Northern Iraq IB environment

#### An Invitation

My name is Tobias Blechschmidt. I am currently enrolled as a postgraduate student at Auckland University of Technology, New Zealand, specializing in International Business and Management. I am looking for volunteers to participate in my study about cross-cultural negotiations.

This research is conducted as part of a Master's of business research project and to obtain a Master qualification. The findings of this research might be used for publication and /or as part of a PhD. Qualification at a later stage.

The purpose of this study is to investigate factors that impact cross-cultural negotiations in the northern Iraqi oil industry. The specific aim is to identify behavioural patterns in negotiation. For this reason I am looking for participants from middle and senior management who operate in this industry and conduct business negotiations regularly to be interviewed for my research.

You have been chosen from a list of service suppliers of oil and gas related services. This list has been compiled and recommended by the domestic Kurdish Ministry of Natural Resources (KMNR). You have been identified because of your business presence in the northern Iraqi oil industry as well as your expertise in cross-cultural negotiations.

The data collection consists of a questionnaire entailing 25 questions and a short interview in which you are asked to reflect on your negotiations that you have conducted in within the last 12 months (please see question sheet attached). The interviews will be audio-taped and then also transcribed for research purposes. Your individual responses will be kept confidential and only the aggregate information will be made public.

Participation in this project is voluntary. I as the researcher will try to answer any questions and address any concerns that you may have at any stage of the interview or data collection. You have the opportunity to withdraw any information that you have provided for this project at any time without prejudice. In the event that you withdraw from the research project all relevant information including tapes and transcripts, or parts thereof, will be destroyed. The identity and privacy of you and all other participants will be treated with care and confidentiality. I assure you of high respect at all times.

Thank you very much for your time and help.

Yours sincerely,

Tobias Blechschmidt

PS: If you have any further questions please feel free to contact me via email or read the following frequently asked questions and answers:

#### Questions that you might liked to have answered:

#### What are the costs for me of participating in this research?

There are no costs for you involved. Participating in the study will take you approximately 30mins of your time.

#### Are there any discomforts and risks for me involved?

Your privacy is always respected. You are not placed at any risk in this research, and you are unlikely to experience any discomfort either.

#### How are discomforts and risks for me alleviated?

If you feel that questions are asked that you are not feeling comfortable to answer you may decline to answer any question at any time. You can also ask for the audio-tape to be turned off, have a response deleted, or choose to have the interview terminated all together. I will make it my priority to treat your concerns with utmost care and respect.

## What do I do if I want to participate in the study?

Please inform myself as the researcher of your decision within the next 14 days by email; the email address is provided below.

## Do I have to give written consent to participate in this research?

All participants who agree to take part in the research have to provide their written consent. You can do this either by email: please sign and send the signed attached consent form. You can also chose to sign the form at the interview.

#### Will I receive feedback on the results of this research?

All participants who are interested in the findings of this research will receive a summary of the findings. If you are interested in the findings please tick appropriate box on consent form.

# What do I do if I have concerns about this research?

Any concerns regarding the nature of this project should be raised with me.

You can also contact the Project Supervisor, Dr. Sabina Jaeger, <sabina.jaeger@aut.ac.nz>, +64 9 921 9999 ext. 5907

Concerns regarding the ethical conduct of the research should be directed to the Executive Secretary, AUTEC, Dr Rosemary Godbold, *rosemary.godbold@aut.ac.nz*, 0064-921 9999 ext. 6902.

# If I want additional information or who should I contact for further details about this research?

Please contact me directly at Tobias Blechschmidt <tblechschmidt@hotmail.com>

#### **Project Supervisor Contact Details:**

Dr. Sabina Jaeger, AUT University; Faculty of Business and Law; Auckland - New Zealand

Phone +64 9 921 9999 ext 5907 Email: <sabina.jaeger@aut.ac.nz>

Approved by the Auckland University of Technology Ethics Committee on type the date final ethics approval was granted, AUTEC Reference number type the reference number.

# **Appendix B: Consent Form**





Project title:

Date:

"Managing cross-cultural negotiations in the current Northern Iraq IB environment."

Project Supervisor: Dr. Sabina Jaeger Research Student - Tobias Blechschmidt Researcher: I have read and understood the information provided about this research project in 0 the Information Sheet dated 10/04/2013. 0 I have had an opportunity to ask questions and to have them answered. 0 I understand that notes will be taken during the interviews and that they will also be audio-taped and transcribed. I understand that collected data might be used at a later stage for academic purposes 0 journal article, PhD. Thesis. e.g. I understand that I may withdraw myself or any information that I have provided for 0 this project at any time prior to completion of data collection, without being disadvantaged in any way. 0 If I withdraw, I understand that all relevant information including tapes and transcripts, or parts thereof, will be destroyed. 0 I agree to take part in this research. 0 I wish to receive a copy of the report from the research (please tick one): YesO NoO Participant's signature: Participant's name: Participant's Contact Details (if appropriate): .....

# **Appendix C: Interpersonal Conflict Questionnaire**

When negotiating:		Very		Neither Likely nor		Very
		Unlikely	Unlikely	Unlikely	Likely	Likely
1.	I usually come up with a plan to influence the negotiation to go my way.					
2.	I usually explain myself well to make sure that the negotiation stays friendly and comfortable.					
3.	I usually go out of my way to make sure that the outcome for the other party is fair.					
4.	I usually make sure that I and the other party can get what we want from the negotiation.					
5.	If something needs to be negotiated, I usually volunteer to do it.					
6.	I usually compromise in order to get something in return from the person I'm negotiating with.					
7.	If the negotiation is not going in my favor, I withdraw from the negotiation.					
8.	I usually come up with creative solutions that allow both me and the other party to get what we want from the negotiation.					
9.	If it seems important for the other person to have the upper hand, I'll give in to them.					
10.	I usually avoid difficult issues to keep the negotiation from getting unpleasant.					
11.	If the other party compromises their position, I'm likely to compromise my position in return.					
12.	It is important to me to communicate and understand the needs of both parties in order to achieve a satisfying outcome for all people involved.					
13.	I usually present information, when negotiating, even if it doesn't necessarily always support my position.					
14.	I usually suggest a solution that allows both parties to meet in the middle.					
15.	I belief it is important to understand the view of the other parties and accommodate their needs.					

Rate your level of agreement with each of these statements.	Strongly Agree	Agree	Neither Agree nor Disagree	Disagree	Strongly Disagree
I hold the view that when negotiating, both sides have to give something up to get rewarded.					
17. What's good for me is really all that matters when negotiating.					
18. I usually avoid engaging in negotiation.					
19. I hold the view that when negotiating, someone wins and someone has to lose.					
20. The feelings of the other party I'm negotiating with are of importance to me.					
21. When negotiating the secret to success is to focus is on common points rather than differences.					
22. My negotiation approach can be aggressive when it benefits me.					
23. I belief compromising in a negotiation is equivalent to losing.					
24. It is not important to me if the other party does not benefit from the negotiation.					
25. When negotiating it is important to me to achieve a good comfort level for all parties involved.					

# Appendix D: Open ended interview questions

# **Indicative Questions for Interview:**

"Participant tells the interviewer about an important negotiation process he or she was confronted with within the last 12 months."

- **Q1.** Can you please tell me about how you conduct a negotiation? What are the secrets of successful negotiation?
- **Q2.** Can you please tell me, when negotiating how important is protocol for you?
- **Q3.** Can you please tell me during your negotiations, how do you communicate with your opposite?
- **Q4.** Can you please tell me what value time has for you when negotiating? E.g. a limited resource that must not be wasted or were you interested in getting to know the other counterpart?
- **Q5.** Can you please tell about the risk that was related with the outcome of the negotiation? Were you willing to take a risk?
- **Q6.**Can you please tell me, did you involve other colleagues or employees in the decision making process?
- **Q7.** Can you please tell is agreement detail-oriented and with specific provisions or open to changes?
- **Q8.** Are you satisfied with the outcome?

## **Appendix E: Ethics Approval**



24 May 2013

Sabina Jaeger Faculty of Business and Law

Dear Sabina

Re Ethics Application: 13/94 Managing cross-cultural negotiations in the current Northern Iraq IB environment.

Thank you for providing evidence as requested, which satisfies the points raised by the AUT University Ethics Committee (AUTEC).

Your ethics application has been approved for three years until 23 May 2016.

As part of the ethics approval process, you are required to submit the following to AUTEC:

- A brief annual progress report using form EA2, which is available online through <a href="http://www.aut.ac.nz/researchethics">http://www.aut.ac.nz/researchethics</a>. When necessary this form may also be used to request an extension of the approval at least one month prior to its expiry on 23 May 2016;
- A brief report on the status of the project using form EA3, which is available online through <a href="http://www.aut.ac.nz/researchethics">http://www.aut.ac.nz/researchethics</a>. This report is to be submitted either when the approval expires on 23 May 2016 or on completion of the project.

It is a condition of approval that AUTEC is notified of any adverse events or if the research does not commence. AUTEC approval needs to be sought for any alteration to the research, including any alteration of or addition to any documents that are provided to participants. You are responsible for ensuring that research undertaken under this approval occurs within the parameters outlined in the approved application.

AUTEC grants ethical approval only. If you require management approval from an institution or organisation for your research, then you will need to obtain this. If your research is undertaken within a jurisdiction outside New Zealand, you will need to make the arrangements necessary to meet the legal and ethical requirements that apply there.

To enable us to provide you with efficient service, please use the application number and study title in all correspondence with us. If you have any enquiries about this application, or anything else, please do contact us at <a href="mailto:ethics@aut.ac.nz">ethics@aut.ac.nz</a>.

All the very best with your research,

Madeline Banda
Acting Executive Secretary
Auckland University of Technology Ethics Committee

Cc: Tobias Blechschmidt tblechschmidt@hotmail.com; dgy3502@aut.ac.nz